THE GLACIAL GEOMORPHOLOGY OF WEST-CENTRAL NEWFOUNDLAND: HALLS BAY TO THE TOPSAILS

## CENTRE FOR NEWFOUNDLAND STUDIES

# TOTAL OF 10 PAGES ONLY MAY BE XEROXED

24

(Without Author's Permission)

CHRISTOPHER M. TUCKER



373535 ELIRE FOR NELD. STUDIE 1AR 1 0 1975 NEWFOUND



THE GLACIAL GEOMORPHOLOGY OF WEST-CENTRAL

THE TOPSAILS

C) Christopher M. Tucker

A thesis submitted in partial fulfilment of the requirements

for the degree of Master of Science.

Department of Geography Memorial University of Newfoundland St. John's, Newfoundland



THE GLACIAL GEOMORPHOLOGY OF WEST-CENTRAL NEWFOUNDLAND; HALLS BAY TO

ABSTRAC

THE TOPSALLS

Christopher M. Tucker

During the late Wisconsin glaciation, ice flow in west-central Newfoundland was to the north-northesat and northesat, sub-parallel to structural lineations. Constructional forms at this stage included drumling, drumlinoid forms and ribbad morgine.

The coast of Halis Bay was deplaciated about 12,000 B.P. in a relatively short period during which glacionarine deltas were formed at Springdale, Dock Point, White Point, Barney's Brook, West Pond, South Brook and Sugarloaf; the latter three being remants of a continuous terrace. Subsequent net isostatic and substitic change positioned the deltas approximately 250 feet (75 meters) above present pes level.

After the initial coastal stage of deglaciation, ics withdrew inland by stagnating in the valleys and lowlands leaving ridged ablation moraine and kettle topography.

Ice recoded in this manner to a plateau level 19 miles (30 kms.) from the coast, where a pause in retreat occurred. During this stage a series of recessional moraines was built and a zone of eakers formed. near Barney's Brook and southeast of Sheffield Hill. Surficial crevasse fillings are also found within this zone. A final, topographically controlled flow into the Kitty's Brook - Chain Lakes valley system followed by stagnation, resulted in the fabrication of a series of recossional-ablation moraines.

Last ice in the field area was situated to the southwest of Gaff Topsail and is evidenced by an area of hummocky, disintegration " moraine.

## TABLE OF CONTENTS

Pag

53

73

96

102

111 0

1.21

126

131

List of Figures List of Tables List of Plates

Acknowledgements -

INTRODUCTION

CHAPTERS

1. THE FIELD AREA

2. THE HALLS BAY DELTA SYSTEMS

3. ESKERS AND CREVASSE FILLINGS

4. THE KITTY'S BROOK - CHAIN LAKES NORATINES

5. GLACIAL MAP COMMENTARY

6. CONCLUSIONS AND DISCUSSION

Selected Bibliography

Appendix 1 Techniques, Instruments and Analyses

2 Textural Analyses Synthesis

Abbreviations

## LIST OF FIGURES

### Figures

2-3

2-41

3-1

3-2

topography and brainage - initiatiated isoline map	1-1	Topography	and	Drainage -	Illuminated	Isoline May
--	-----	------------	-----	------------	-------------	-------------

1-2 Bedrock Geology

2-1 Springdale Delta

2-2 West Hall's Bay; Deltas and Ice Fronts

Cumulative Curves, Indian Brook Valley Sediments, Frequency Curves, Indian Brook Valley Sediments

2-5a X-Ray Pattern, 2-30° 29 (theta) (Untreated)

2-6 Lower Halls Bay; Deltas and Ice Fronts

Eskers and Crevasse Fillings; Locations and Sample Points 54 Cumulative Curves, Barney's Brook and Sheffield Hill

Cumulative Curves, Sarney's Brook and Sherrield Hill Deposits

3-3 Frequency Curves, Barney's Brook and Sheffield Hill , Deposits

4-1 Recessional-Ablation Moraine; Kitty's Brook - Chain Lakes

4-2 Cumulative Curves, Chain Lakes Samples

4-3. Frequency Curves, Chain Lakes Samples

4-4a,b Till Fabric Diagrams

c,d Till Fabric Diagrams

4-5a,b Rose Diagrams

c',d Rose Diagrams

5-1 Glacial Map

See Back Cover

Page

22

29

32

33

41

65

66

77

80

81

84

	사람이 있는 것 같은 것 같은 것 같은 것 같은 것 같은 것 같이 있다.	
6-1	Recessional Ice Positions and Ice Flow Directions	10
6-2	Locations Mentioned Outside the Field Area	10
A-1	Mean vs Skewness	12
A-2	Skewness vs Standard Deviation	. 12

A-3 Skewness vs Kurtosis

•••

VS AUTOS

#### LIST OF TABLES

#### Table:

4-2

1-1 Climatological Records for Buchans and Springdale

Page

15

31

52

57

67

69

2-1 Textural Analysis; Calturated Data for Indian Brook Valley Samples 72 M-5 4772 M-8

2-2 Elevations of Terraces and Raised Shoreline Features on the Halls Bay Delta Systems

3-1 Distinctive Features of Eskers and Crevasse Fillings

3-2 Textural Analysis, Calculated Data for High Central -Plateau Samples 72 E-1 - 72 E-4

 3-3 Textural Boundary (φ units) at 90% of Cumulative Weight
4-1 Textural Analysis, Calculated Data for Chain Lekes Samoles 72-2 - 72-3

Till Fabric Data, Calculated and Interpreted

#### LIST OF PLATES

## Plate 1. The Fiel

1. The Field Area, E.R.T.S. Image

2. The Springdale Delta, Southwest Scarp

3. The Springdale Delta, East Scarp

4. Location 3, Zone of Sands and Gravels, Springdele Delta

Page

21

23

23

25

25 28

:28

42

42

45

50

50

59

59

62

68

74

78

78

5. Location 6, Poreset Bedding, Springdale Delta

6. Location 7, Indurated Silt, Springdale Delta

7. Calcite Deposit

8. Location 4, High Discharge Material, Springdale. Delta

9. Bedded Silts and Clays, Indian Brook

10. The West Pond Terrace

11. The Sugarloaf Terrace

12. The Southbrook Terrace

13. Shell Sample 72-295

14. Lower Halls Bay

15, The Barney's Brook Esker

16. Perched Erratics, Barney's Brook Esker

17. The Sheffield Hill Esker

18. The Foretopsail Area

19. Sample Pit 72 E-4

20. The Kitty's Brook - Chain Lakes Area

21. The Chain Lakes Moraines

22. Moraine .72 -3

23. C.N.R. Mile 343, The Kitty's Brook Delta

24. The Kitty's Brook Delta

Page

92.

93

25. Kitty's Brook Valley

Plate

#### CKNOWLEDGEMENTS

The writer wishes to thank several individuals and organizations who sided in the preparation and writing of this thesis.

Mr. R. J. Rogerson, Department of Geography, Memorial University of Newfoundland, supervised the research and writing of the thesis and provided advice that was of benefit at all stages of preparation,

Dr. D. R. Grant of the Geological Survey of Canada in Ottawa supplied logistical support early in the field season (through a D.R.E.E. - D.M.A.R. glacial geology-geochemical survey), and advice during the preliminary thesis preparation including permission to use radiocarbon date 6.5.C.-1733.

Dr. J. B. Macyherson, Department of Geography, Memorial. University of Newfoundland, gave counsel in the initial planning stages of the thesis and ongoing discussion throughout the study.

Dr. M. Slatt, Department of Geology, Memorial University of Kewfoundland, advised the writer on sedimentological techniques and made laboratory facilities available for the analyses.

Kr. M. Crane, Department of Geography, Memorial University of Newfoundland, advised in the preparation of diagrams and aided in the reproduction of the glacial map.

Ms. C. Deorksen contributed her excellent sketch of the Foretopsail which appears as the frontispiece of the thesis.

Newfoundland and Labrador Forestry Service provided facilities including airphotos for an initial enalysis of the terrain.

Canadian National Railways arranged transportation in the

For these and other abbreviations, see Appendix 3, p. 131.

mouthern sections of the area, without which the field work involved in Chapters III and IV would have been difficult to accomplish.

à.

#### INTRODUCTION

#### **Objectives**

The raised Pleistocene deltas at Mails kay and the inland glacial features of west-central Mewfoundland have been described by several authors in the past fifty years. Auch of the early likerature is qualitative in its approach and tends only to identify problem areas and speculate on solutions. A detailed analysts of the morphology and chronology of the various deposites is required to complement related studies of the north coast (Grant, 1969, 1970) and west coast (Brookes, 1969) and to deternink in particular whether the inner-outer drift zones described by Jennies (1960) in eastern -Newfoundland and by Lundquist (1965) in the field area do in fact exist. Lundquist also raises the possibility of invasion of the Halls hay area by early Wisconsin Labridor ice, which he bases on evidence from the Jenness 1960 inchese map and southeast-ordered strike.

The present study involves an overall air photo analysis and the production of a giscial map plus the detailed analysis of three specific problem areas defined by previous literature and the preliminary air photo interpretation. It is hoped that a more rigorous approach this formerly attempted will provide some specific answere, or will at least re-define problems with new information.

#### Location of Study Area

The field area of 624 square miles (1597 square kms.) is located in west-central Newfoundland and extends from Springdale in the north to



## PLATE 1. THE FIELD AREA

- 1. Halls Bay
- 2. Springdale
- 3. Town of South Brook
- 4. Howley
- 5. Burnt Berry Pond
- 6. West Pond
- 7. South Pond
- 8. Chain Lakes
- 9. Little Sandy Pond
- 10. Gull Pond
- 11. Indian Brook

For completeridentifications refer to the Glacial Map, Chapter V. The light area between Chain Lakes and Little Sandy Pond shows a recent snowfall on the High Central Plateau.

SCALE: 1 CM. = 10 KMS. 1 INCH = 16 MILES

EARTH RESOURCES TECHNOLOGY SATELITE IMAGE 5, 11, 72 just south of the C. N. R. line at the Topsails on the High Central Plateau (Plate 1). Twenboid and MacClintock (1940) identify the plateau as "that part of the Atlantic Upland between the Grand Lake -White Bay Basin and a line extending from Hall[s] Bay to the head of Red Indian Lake" (p. 1702).

To the southwest, a section of the upper Kitty's Brook and Chain Lakes drainage system is included in the field area.

#### Organization of the Thesis

Chapter I serves as an introduction to the field area; the topography, general bedrock geology, drainage; climate, vegetation, access and human geography of west-central Newfoundland are discussed briefly.

Chapter II describes the Springdale and lowar Halls Bay delts systems. It exemines the various unconsolidated deposits in Indian Brook valley and several marine shell samples found in the area. A description of the depleciation process and chronology of the coastal section of the field area is tendered.

Chapter III is concerned with a zone of esker-like features in the south-central portion of the study area that ware identified by Lundquist (1965) as open crevases fillings and " of limited value to interpreting ice flow: " (p. 296). A morphological and textural analysis was conducted on specific deposits and a reinterpretation based on these results is presented.

Chapter IV examines a series of ridges located in the southwest corner of the field area at Kitty's Brook and Chain Lakes. The features were briefly described by NacOlintock and Twenhofel (1940) as being

recessional moraines. They are a significant part of the overall study since it is deduced that last ice melted in this vicinity and, as such, the deposits might indicate a late, topographically controlled ice flow. Till fabric and textural analyses were completed and a method of formation is suggested for the deposits.

Chapter V presents a glacial map of the area based on a detailed interpretation of 1:15,840 and 1:50,000 air photos. An explanation of the legendmis included and it is hoped that this, in conjunction with the map,will be of some practical value in future glacial chronologies and glacial drift studies for mineral exploration.

Chapter VI summarizes the work presented in the previous chapters and synchosizes a chronology of the last stages of Misconsin glaciation and the sequence of deglacistion. Several comments are made pertaining to previous studies and a note is made of possibilities for future research within the field area and in tracts peripheral to it.

Techniques, Field Instruments and Analyses are discussed in Appendix 1. They outline the basic procedures used in analyzing sediment characteristics and landforms discussed in the various chapters.

A brief second appendix presents graphs of the unconsolidated, sediment data plotted in several texturally descriptive combinations,

#### Logistics

1. 18

The initial part of the field searon (May) was spent in Springdale and around the lower Halls Fay area. Travel was by car while foot traverses were completed to less accessible areas. Camp was established mear Riverbead Brook for study around lower Halls Bay and the Department of Mines base camp was used for a period of about three weeks while in Springdale and vicinity. The writer was employed for a period during the field season with a Department of Regional Economic Expansion - Department of Mines, Agriculture and Resources, glacial drift project under the direction of Dr. D. R. Grant of the Geological Survey of Canada, (Grant 1973).

The second part of the field season involved work on the High Central Plateau. Access to the Kitty's Brook - Chain Lakes area was by C.N.R. "speeder" from Howley; camp was set up near the southern end of Chain Lakes and foot traverses were complated over a tadius of about three miles. Various stops were made along Kitty's Brook valley on the return journey.

The Topsails area was approached from the east. Again a C.N.R. "speeder", this time from Millertown Junction, was the initial mode of traduport. From Gaff Topsail, the area was powered extensively on foot with a base camp being established at Foretopsail from which traverses were made to Barney's Brook and Sheffield Hill (Chapter IV).

A final foray was made around Springdele, Halls Bay and the major valleys in the northern part of the field area. All secondary roads ware driven to field check the sir photo analysis. Several of the private roads, that is, to Gull Bridge Mines and the Price (Newfoundland) Pulp and Paper Ltd. road to Dawes Pond and Lake Bond were also covered to check interpretations further.

## CHAPTER I

#### THE FIELD AREA

### Topography

The field area (Fig. 1-1 and Plate 1) may be generally divided into two physiographic regions; (1) the Halls May - Gull Pond lowLands, (2) the High Central Plateau.

 The Halls Bay - Gull Pond area is characterized by a northnortheasterly structural-topographic trend which is emphasized by glacially accured bedrock ridges. Halls Bay proper is a fjord with a 130 fachema (270 meters) threshold close to shore. In this morthern section of the field area, summit elevations along from 750 feet (225 meters) southwest of Gull Pond to about 500 feet (150 meters) just morth of Springdale. Prominent bedrock peaks occur at 456 feet (137 meters) at Springdale and 855 feet (257 meters) at Rowsell Hill behind the town of South Brook.

Neale and Mash (1963) map a tight synclinal axis following the western shore of West Fond is a northeasterly direction to the mouth of Indian Brook. Within the central section of the Halls Bay- Gull Fond lowinds, the authors describe a broad synclinal trough with its aris roughly paralleling the upper reaches of West Brook. The synclinal folds reach altitudes of 800 feet (240 meters) northeast of Sheffield Lake and 974 feet (292 meters) at Nutaeg Hill northeast of Gull Fond. Elevations of 500 feet (150 meters) are typical of the axial area of the trough.



2. The second topographic subdivision includes the area south of Gull Pond in the east, Sheffield Lake in the west and extends beyond the southern limit of the field area. This is the High Central Plateau as described by Twenhofal and MacClintock (1940). The boundary between the two physiographic units is sharply defined by a break in slope at the 900 feet (240 meters) contour. Solifiuction lobes and thick drift cover the break in slope. Flateau elevations alone gently from 1500 feet (450 meters) in the south to 1000 feet (300 meters) at the northern edge of the plateau. West-east till from 1500 feet (450 meters) meer Chain Lakes to 800 feet (240 meters) at Little Sandy Pond. The terrain between Little Sandy Pond and the eastern border of the field area may be included in the first physiographic modivision.

The Topsails are prominent hills on the High Central Plateau which are referred to as monadnocks by Twenhofel and MacClintock (1940). The authors propose that before glaciation the ridges were probably "high places," with graded slopes, on the plateau, and that jointing in the granite bedrock facilitated plucking which led to their becoming crass. In view of their description of the Topsails being formed by glacial action rather than by long term circumdenudation, it seems rather inappropriate that they refer to the peaks as monadnocks. The Topsails are all elongated 040-045 and rise 200-230 feet (60-75 meterm) above the surrounding topography. Sheffield Hill may be included with the granite Topsails though it is located further west. Elevations of the peaks are us follows: Sheffield Hill, 1639 feet (492 meters); Foretopail Hill, 1610 feet (463 meters); Main Topsail, 1822 feet (547 meters); and Gall Topsail, 1650 feet (495 meters).

### General Bedrock Geology

The bedrock geology of the field area has been mapped in part by Kallickoski (1953) and more completely by Mesle and Mash (1963). The following generalizations have been made from Neale and Nash; numbers on Figure 1-2 indicate bedrock type and are identified in the text.

The High Central Plateau is underlain by Devonian Topsails granite (18); pale red equigranular granite, quartz, monzonite and granodiorite. The central synclinal troughs including the area as far north as Springdale consist of the Springdale Gubup (15), that is, zones of (a) red sandstone and conglomerate, red and greenish grey shale and minor limestone, (b) red sandstone, fongiomerate, limy. siltstone and shale, (c) silicic flow and pyroclastic rocks and (d) basic flow and pyroclastic rocks. The Ordovician Lush's Bight Group (5), located north of Springdale, is formed of schistose basalt and andesite, minor pyroclastic rocks, greywacke, slate and chert. The area around Gull Pond is part of the Ordovician Exploits Group (6); the Roberts Arm Formation includes basalt, minor pyroclastic rocks, silicic flow rocks and basic sills. The Crescent Lake Formation contains shale, minor chert, greywacke and rhyolite. The area between Little Sandy Pond and Halls Bay comprises various patches of granite. granodiorite and syenite similar in age to the Topsails granite (18), as well as Devonian quartz diorite, diorite and gabbro (17). Southwest of Sheffield Lake there is a zone of Silurian andesite, basalt flow and pyroclastic rocks (11).

The field area is part of the Central Mineral Belt of

.9

Number of rock type

5...

11

17

Legend

Ordovician - Lush's Bight group. Ordovician - Exploits group Silurian andesite

Group or general description

Devonian diorite

Devonian Topsails granite



Newfoundiand, (@see et al., 1970), and as such has been subject to extensive mineralogical exploration by the American Smelting and Refining Company in the south and British Newfoundiand Exploration Limited and Noranda Exploration towards the northeast. While a number of shortlivid mining operations were located in the Belt, Guilbridge Mines at Gull fond was the only one within the field area. Neale and Nash (1963) state that a headframe and mining plant were first constructed in 1956 but after an initial run, production ceased in 1971. Rose et al. (1970) note that the occurrence of suifide minerals on Gull Fond has been known since 1905. Pyrrhotite, pyrits, chalcopyrite and magnetife as well as quartz, chlorite, sericite, biotite, cordierite and tremolite-scingolite occur in lenticular masses in sheared zones of the Roberta Arm Group. Many of the above minerals are found in erratic boulders of unknown provenance and have been studied by Boliden Mining Company of Sweden in an attempt to discover their specific origin.

#### Drainage

For the most part drainage in the central and morthern sections of the field area is to the northeast in drift-filled valleys following structural lows (Fig. 1-1 and Fiste 1). Streams are not well incised in their courses except for Indian Brook which is deeply entrenched and flows over bedrock in its lower sections. Between the T.C.H. and Springdale, Reander scars are visible over a broad terriced. These however may relate to a much larger deglacial river flowing within Indian Brook valley, and graded to a higher sen lavel.

On the High Central Plateau the drainage pattern is deranged (typical of glaciated old plateaus) though a haphazard northeast trend

. 11

predominates across the bouldery surface. Near the southwest corner of the plateau, flow is southwest down Kitty's Brook valley and to the northwest in the upper Chain Lakes valley to Sheffield Lake, 'A drainage divide from which water flows moutheast to Red Indian Lake is located just mouth of the field area.

Although ponds are numerous throughout the field area, most are small, less than 2 square miles (5 sq. kms.) in area. The largest, West Pond 3.4 square miles (8.8 sq. kms.) and South Pond 3.6 square miles (9.1 sq. kms.) are confined to deep valleys with areas of thick till and glaciofluvial material. To the west and east, Sheffield Lake and Gull Pond are located in structural basins. The Chain Lakes system, at the southwest corner of the area, is demined by a sories of worstand deposite (see Chapter IV).

#### Vegetation

The vegetation of west-central Newfoundland can be related to physiographic subdivisions. The plateau level is devoid of forest cover; vegetation consists of alder thickets (Migue) and occasional 'patches of stunted white spruce (*Piesz glauca*) in poorly drained, sheltered areas. 'Generally, plant species are limited to low bushes on the morainal cover and grasses on the numerous string bogs. On the terrain north and below the High Central Plateau, where drainage is well defined and soils more fully developed, forest cover is extensive.'

Durman (1964) states that balasm fir (*Abless balasmea*) occupies all soile except the extremely dry and vet ones in undisturbed forests of central Newfoundiand. These are covered with either black

spruce (Fiesa mariana) or alder. Hardwood forests are always of fire origin and contain white birch, (*Betula papyrifera* March) treabling aspen, (*Populus tremuloides*) and pin cherry (*Promus permuluaniaa*). Elack spruce is also a dominant burn species and is prominent around Halls Bay.

#### Climate

Obtaining long term elimatic data for the field area in difficult since no stations have been in operation for extended periods, on this section of the island. Long term statistics from Buchans (50 miles (80 kms.) southwest of Springdale) are used as being indicative of climatic conditions on the High Central Plateau and monthly averages for Springdale (1971) describe coastal conditions (Table 1-1). While the two groups of figures are not strictly comparable, they do auggest veriations in climate batween the two main zones of the field area.

Precipitation is well distributed throughout the year with a slight maximum in the fall: November generally being the wettest month, (Hare, 1952). Winters are relatively mild, though colder on the inland plateau. Snowfall is moderately heavy over the whole field area with more than 100 inches (254 cms.) being experienced on average: Total annual precipitation is slightly higher at Buchand, 36.7 inches (91.6° cms.) compared with 34.7 inches (90.8 cms.) at Springdale. August is alightly warmer than July at the cost but inland the reverse is true. This is probably a result of lingering Arctic ice and the Labrador Current which delays or moderates summer warmth on most of the costal sections of the faland. Hare (1952) states that the vegetative season starts between the 15th-20th of May, more than a month behind Kontreal or Ottawa. The frost free season is about 100 days, from early June to mid September, in the southern part of the field area; in the constal zone the season is probably longer, again due to the modifying marine : influence.

#### Access and Ruman Geography

Access to the perimeters of the field area is excellent. The Trans Ganada Highway follows the east termination and northerly hargins and the Ganadian National Bailway cuts the southern boundary; access to the plateau is available from either Bowley in the west or Millertoym Junction in the east. Both mattlements were used as starting points on traverses to the High Gentral Plateau.

Springdale, the largest town in the field area and the economic hub of Green Eay, is located 73 miles (117 kmm) west of Grand Falls and 118 miles (139 kms.) east of Gorner Brook by road. Until recently, Springdale was a dormitory town for a population working at Whalesback and Little Bay mines which have since beam closed. Employment in the town is generated by service satablishments for the populus of Green Bay, a provincially operated Vocational School, a D.M.A.R. geochemical aboratory and several diamond drilling operations. Much of the population is employed in the woods operations of Frice (Newfoundlaod) Fulp and Paper and Bowaters, Ltd. both of which have extensive cutting rights in the field area. Springdale also provides a link with Notre-Dame Bay and Labrador on the C.N.R. coastal bat service.

## TABLE 1-1

## CLIMATOLOGICAL RECORDS FOR BUCHANS AND SPRINGDALE

Buchans Long Term (19	42-197	1) Clim	stologi	cal Rec	ord		ж се				· · ·	300 A 20
<b>1</b>	J	F	M	A	M	T	Ţ	A	• <u>s</u>	· <u>o</u> ·	N	D
Mean Daily Temp °F.	16.9	15.7	21.0	31.5	42.1	52.6	60.0	59.0	51.8	41.5	32.5	22.2
" Max." " "	23.3	22.5	28.1	38.1	50.9	62.5	70.0	67.7	60.4	48.4	38.1	28.3
" Min. " " "	10.5	8.9	13.8	24.8	33.3	42.6	49.9	50.3	43.2	34.5	26.9	16.0
Snowfall (inches)	23.2	21.0	18.1	9,6	1.3	0.9	0	0	0	ī.7	9.9	20.6
TOTAL PPTN.	3.31	2.92	2.35 -	2.10	2.42	2.60	2.94	3.68	3.57	3.33	4.11	3,40
· · ·					100							
Springdale 1971 Clime	tologi	cal Rec	ord									2.2
	ī	F	M	A	м	J	3	A	S	0	N	D
Mean Daily Temp °F.	20.6	. 15.8	30.0	37.9	50.1	53.8	62.0	65.3	52.8	4,24	-35,6	16.1
" Max. " " "	24.8	28.1	37.9	44.6	60,3	63.8	73.3	72.9	62.3	49.6	42.5	25.6
" Min. " "	13.9	13,5	22.1	31,2	39.8	43.8	50.7	57.7	43.3	35.1	28.6	6.5
Snowfall (inches)	39.8	6.5	30.0	8.0	0	0	Q .	0	0	. 0	13.0	19.8
TOTAL PPTN.	4.21	1.46	3.05	3.46	1.65	2.78	2,76	3.97	3.03	2,19	4.04	1.13

South Brook, located 15 miles (24 kms.) south of Springdale was a shipping center for copper concentrate from the now defunct Gulbridge Mines. Its economic viability is derived from the logging operations of private contractors for the paper wills at Grand Falls and Corner Brook.

The Halls Bay area is presently, undergoing some small economic revitalization with the development of its natural tourist facilities. Both provincial and private campaites have been opened during the past few years.

### CHAPTER II

#### THE HALLS BAY DELTA SYSTEMS

## Introduction

Halls Bay contains a number of reised Fleistocene deltas at various locales from Springdale in the north to South Brock and West Bottom at the head of Halls Bay. The deltas have been described by Lundqvist (1965) who proposes a general scheme for construction of the system following the Jannese (1960) theory of an inner-outer drift. zones. He hypothesizes individual ice lobes protruding from an inland ice sheet and filling the valleys which feed into Halls Bay. Jenness argues that the inner-outer drift zones in northeast Newfoundland resulted by either a still-stand at a position of discontinuous end moraine or a final major advance from farther inland. He identifies the outer drift zone (coastal wide) by ground moraine and the inner drift.zone 'by "slightly younger eskers, kames mid ground moraine" (p. 161), Supposedly, final matting produced the glaciofluvial deposite the radiate coastward to for deltes.

Despite the Jencess argument, Lundqvist mentions that no end moraine could be distinguished at the some boundary in vest-central Newfoundiand. To substantiate the theory he describes the inner-outer boundary on the basis of thick, almost continuous, inner drift and thin outer drift where-". . . bedrock structure is clearly visible through the thin overburden, even where bedrock is not exposed" (p. 299). The deltas at Springdale and South Brock, as well as Kings Fohn and Botwood (the latter two being located outside the field area) are interpreted by Lundqvist as being contemporaneous and part of the boundary between inner and outer drift, since such a halt would allow the deposition of much thicker glaciofluvial sediments than would a continuous recession. of the ice. The possibility of the deltas being deposited in ice-dammed waterbodies as described by Jenness is also suggested by Lundqvist.

Jenness proposes a series of isobases from elevations of former marine lawels,based on his work on the northeast coast and that of Flint (1940) for the west coast. He assumes contemporaneous deglaciation for the whole north coast, although unproven, and enviseges a O isobase of no mat postglacial emergence passing through the isthmus of Avalon, 100 feet (30 maters) of upliff at Eastport in Bonavista and 250 feet (75 meters) at Kalls Eay relative to present sea level. Since rebound is proportional to ice thickness, the values indicate late Macomain ice compentrated over the west-central section of the island.

A glacial study completed by Damman (in Wilton, 1957) portrays the Halls Ray delta deposits as being similar in origin to those of St. Georges Bay (MacClintock and Twenhofel (1940) and Brookes (1969)). The process as presented in the Wilton report would have an extensive glaciofluvial period with a readvance of ice depositing till 1-2 fact (0.3-0.6 useters) thick on top of the deltas. Damman continues by hypothesizing a gradual isostatic rise of the land which resulted in rivers flowing into Halls Bay outting their way through old deltaic deposits, leaving some parts of the deltas as high terraces.

Several authors have studied the Halls Bay deltas as individual units within the system. MacClintock and Twenhofel (1940) describe the Springdale terrace as an emerged Pleistoceme delta of samd and

18.4

gravel that has scattered boulders stream on top at 260 feet (78 meters) : a.s.1. They depict the delta as being over 1500 feet (450 meters) long, 800,900 feet (240-270 meters) wide, about 200 feet (60 meters) thick and resting against a bedrock hill northwest of Springdale which is situated on a 30 feet (9 meters) terrace at the foot of the main scarp. The authors state that the deposit is an eroded remnant of a once extensive deposit of much-pitted outwash that filled all the valleys draining into Halls Bay and that terraces in the South Brook valley at 15, 55 and 240 feet (4.5, 16.5 and 72 meters) are fragmented remains of the continuous "Halls Bay delta." The lower two terraces are wave-cut in drift while the third is the original depositional surface of the delta.

The Springials delta is also described by Lundqvist (1965). Be suggests it was formed, particularly in its eastern section, by meltwater flowing south, noting that the bade dip slightly away from the hills northwest of the town. Similarly, bedding in other deposits in the area slopes outward from local centers. Unlike MacClintock and Twenhofel, Lundqvist visualizes the 240 feet (72 meters) terrace at South Brook as being an isolated terrace formed in front of West Brook Alley.

Since theories and morphological descriptions strong for the Halls Bay deposits, an attempt has been made to quantify and qualify the characteristics of the system and present some conclusions as to their method of formation.

#### THE SPRINGDALE DELTA

#### Morphology

In actual size the Springdale delta is somewhat larger than
described by MacClintock and Twenhofel (1940). On its eastern face the delta is over 3600 feet (1080 meters) long, while the southwest scarp (Plate 2) is 2500 feet (750 meters) in length from the bedrock cliff northwest of the town to its southern spit-like extremity.

The highest point on the delts surface was measured at 252 feet (75.6 meters) a.s.l. The surface of the deposit is harmocky and shows some dissection. Near the eastern edge of the scarp there is a kettle remnant at 200 feet (60 meters).

Several terraces and raised shoralines are visible on the eastern face of the delta (Plate 3). A weakly developed terrace is visible, which varies in elevation from 209 feet (62.7 meters) at the top southern end, to 220 feet (66 meters) further east. Towards the extreme southeast and of the delta a reading of 181 feet (54.3 maters) was obtained on a raised shoreline, while further north mear Huxter's Brook a height of 150 feet (45 meters) was measured at the edge of the scarp. This variation in height may be a result of erosion by the brook at location 2 (Fagure 2-1).

As praviously mantioned, the town of Springdale is built on a lower terrace of the delta though it is approximately 50 feet (15 meters) rather than 30 feet (9 meters) a.s.1. as reported by MacOlintock and Ymenhofel (1940). Near the government wharf, elevations of 35-40 feet (10.5-12 meters) are probably a result of erosion and slump elong Muxter's Brook. A break in slope occurs at.58 feet (20.4 meters) on the east side of the delta and at 89 feet (26.7, meters) in the gravel pit mear the Department of Highways Depot. (location 3, Plate 4).



PLATE 2. THE SPRINGDALE DELTA, SOUTHWEST SCARP.





PLATE 3. THE SPRINGDALE DELTA, EAST SCARP.



PLATE 4. LOCATION 3, ZONE OF SANDS AND GRAVELS, SPRINGDALE DELTA.

# Sedimentary Structure

Exposed bedding in the delta is chaotic; however, descriptions are made with reference to various plates (especially Plate 2) and Figure 2-1.

At location 3 there is a mass of highly dissocted silty material at 78 feet (23.4 meters) on which a strike of OlO'and dip of 10° R was recorded. From the location and mélange of material it would appear that this was a zone of turbidity or deposition of fines related to a period of lower melt-water discharge.

Large bouldery deposits at location 4 are 118 feet (35.4 mstore) a.s.l. and contain well rounded clasts up to 3 feet (0.9 meters) in dismater cemented with a gravel matrix. Thirty feet (9 meters) of foreset material above the boulder zone contains rounded stores 4-3 inches (1.3-7.6 cms.) in diameter and has strike and dip of 010, 10° R.

Near the southwest end of the delta at point 6 (Plate 5) there are excellent exposures of foreset gravels at 130 feet (39 meters). Redding planes, measured at various points, strike 060-070 with dips of  $15^{\circ}-25^{\circ}$  R. Texturally, the deposite are finer than the foresets noted at location 5; the size range being 3-14 inches (1.3-3.8 cms.).

At the southeast corner of the delta at a height of 85 feet (25.5 meters) (location 7, Plate 6) there is a mass of slightly indurated sandy sediment striking 346 and dipping 25° L. While the bedding plane does not correspond to location 3, the deposits may be related in that they represent similar depositional environments.

Bedding northeast of Huxter's Brook is indicative of a flow of material through the stream gap; north of the delta,chaotically bedded



PLATE 5. LOCATION 6, FORESET BEDDING ORIENTED 060-070, 15°-25° R, SPRINGDALE DELTA.



PLATE 6. LOCATION 7, INDURATED SILT, ORIENTED 346, 25° L, SPRINGDALE DELTA.

sands and silts are visible 81 feet (24.3 meters).a.s.1.

Northwest of the above location at an elevation of 178 feet (53.4 meters) a large mass of calcite is visible (Plate 7). D.M.A.R. sources indicate that the calcite is Ordevician and related to the Lush's Bight Group. The height of the exposure corresponds favorably to the previously mentioned 180 feet (54 meters) erosional terrace.

An attempt was made to relate lateral meltwater or kame terrace levels on the north side of Indian Brook valley to deposition of the Springdale delta, however breaks in slope were poorly defined and no specific correlations were possible.

#### Discussion

given the data outlined above and presented in Figure 2-1 it is suggested that the Springdale delta was deposited in Halls Bay from meltwater flowing through the designated route and along what is now Huxter's Brook. The modified kettle hole on the eastern edge of the delta and the bedrock meltwater channel indicate that ice blocked Indian Brook valley during deposition and was close to the weatern face of the delta. Recorded dips and strikes signify that while initial flow was down the north side of Indian Brook valley, meltwater involved in delta deposition also flowed to the east-northeast and deposits wrapped around the massive bedrock outcrops.

Tides and currents in the bay modified the delts and elongated its southern tip into the spit-like shape shown in the plan view (Figure 2-1).

Nolumes of meltwater varied considerably during delta deposition. This is implied from the mass of bouldery material at location 4,

(Plate 8), indicative of a period of very high discharge. Furtherdeposition of large amounts of fines onto the boulder layer would signify, a substantial slowing of meltwater flow,

It is further suggested that the delta was at no time much more. extensive is a north-south direction than it is now. British Admiralty Chart, No. 4591, does not indicate deposition of large quantities of sediment outside Indian Brook Arm, tather the 100 fathoms (180 meters) threshold is close to shore at this point.

G Tee later retreated in Indian Brook valley. During this time the delta was elevated by isostatic rebound, with subsequent 210 Teet (63 meters), 180 feet (34 meters) and 50 feet (15 meters) still-stands occurring after initial delta deposition. No correlations between still-stands and glacial retreat are envisaged, rather the term is interpreted as meaning a point in time when the rates of isostatic and eustatic change were briefly comparable resulting in the cutting. of terraces on the critical delta.

#### Indian Brook Valley Sediments

On traverses along Indian Brook valley, fine silt-clay deposite were noted at various Jocations (Figure 2-2). For the most part the deposite are 1-3 fact (0.3-0.9 meters) thick and are covered by varying amounts of glaciefluvial gravels, however, one deposit 5 miles (8 kms.) west of Springdale and north of the Indian Brook Bridge is more than 41 fact (12.3 meters) thick from its thin glaciofluvial gap to the stream bed. The mediment mass is bedded (Flate 9), and shows varying pink and gray colorations.



PLATE 7. CALCITE DEPOSIT AT NOTCH, 180 FEET (54 METERS) A.S.L. SPRINGDALE. NORTH.

PLATE 8. HIGH DISCHARGE BOULDER ZONE CAPPED BY FORESET GRAVELS, SPRINGDALE DELTA.





Two possibilities exist for the origin of the deposits. 1. The clays may be marine. This would suggest tapid deglaciation and marine transgression up the valley which at that time was isostatically depressed (as was the whole area), followed by considerable glaciciluvial deposition from inland sources with delta construction at the houth of the valley. The level area west of the T.C.H. - Springdale highway intersection would thus be interpreted as alluvial fill over a flat marine sediment base.

2. The sediments may be lacustrine in origin. They would then have been deposited in ponded water at various locales and elevations along the valley system. This need not have occurred in direct association with delta deposition at Springdale but rather as ice retreated up valley, or in any body of postglacial ponded water.

In an attempt to decide the method of deposition, several adalyses were complated. Four simples were chosen; two were obtained from the massive deposit morth of Indian Brook Bridge. Sample 72 N-7 was from a grey silty leas while 72 N-8 seemed visually finar in textors and had a pinkish hus. Two samples were taken from a meander-sear cut, 0.3 mile (0.48 km.) further west, south of the Springdale highway. Sample 72 N-5 was dug from a 6-8 inches (15-20 cms.) thick lens that mloped in an easterly direction and mample 72 N-6 was removed from a 1 foot (0.3 meter) thick lens of sediment 4 feet (1.2 meters) higher and 50 feet (15 meters) further west. By comparing the two sets it was hoped to be able to relate them more clearly; (a) in a total morphological context in Indian Brook valley and (b) as individual sediment amember within "specific deposits."

#### Textural Analysis

Initially wet size and pipette analyses of the sections finer than 40 (phi) were completed as described in Appendix 1. The results were plotted on Cumulative Curves and Frequency Curves (Figures 2-3 and 2-4). The mean, inclusive graphic standard deviation, kurtosis and inclusive graphic skowness were calculated as outlined and the results tabulated in Table 2-1.

Results

#### TABLE 2-1

2				· · · ·
SAMPLE	MEAN	INCLUSIVE CRAPHIC STANDARD DEVIATION	KURTOSIS	INCLUSIVE GRAPHIC . SKEWNESS
72 115	7.73¢	2.21¢ very poorly sorted	0.93 mesokurtic	0.21 fine skewed
72 196	7.28¢	2.690 very poorly sorted	0.87 platykurtic	0.17 near symmetrical
72 M7	4.62¢	1.70¢ poorly sorted	1.67 very leptokurtic	0.52 strongly fine skewed
72 MB	6.70¢	2.06¢ very poorly sorted	1.05 mesokurtic	0.32 strongly fine skewed

#### CALCULATED DATA FOR INDIAN BROOK VALLEY SAMPLES 72 M5 - 72 M8

Samples 72 M-5 and 72 M-6 contain more clays than the final two samples. .72 M-6 has a greater percentage of coarser (3-44) material, thus explaining its near-symmetrical platykurtic curve. Because of its high very fine sand content, sample 72 M-7 is leptokurtic. A secondary mode in the 10-12¢ range has the effect of producing a strongly fine akewed Frequency Curve. All samples have a visible secondary mode in the clay fraction though percentages are considerably higher for samples 72 H-5 and 72 M-6.





v., 7

Figure 2-4

A section was cut from the massive Indian Brook deposit and

the following sequence was measured.

TOP 0.70 inches (2 cmo.) grey 0.60 inches (1.7 cms.) pink 1.85 inches (4.7 cms.) grey 0.90 inches (2.2 cms.) pink 4.10 inches (1.5 cms.) pink 1.9 inches (5 cms.) pink 1.9 inches (5 cms.) grey BOTTOM

### K-Ray Analysis

To test the differences in composition of the various samples, an X-ray shallysis of the sections finer than 80 was attempted, as explained by Carroll (1970) and outlined in Appendix 1. The initial patterns run from 2-30 degrees 20 have been drafted and are presented as Fisure 2-5 a, b, c and d.

# Results

Compositionally the samples were similar. All contained whiorite (144° (001), 74° (002)), amphiboles (8.424°), mica-illite (104° (001), 54° (002)), quarts (3.334°, 4.214°) and feldepare (3.17– 3.194°). No montmovillonite or kaolinite was present. The samples did vary in the amounts of specific minerals they contained. Related to the textural analyses; sample 72 M-5 and 72 M-6 contained greater proportions of clay, thus their patterns are disturbed and have more background "noise." Peake are not as sharp as those of 72 M-7 and 72 M-8 which are pointed and in many cases 3-4 times the height of the background pattern.









### Stereo Microscope Analysis

A stereo microscope analysis was completed on samples 72 M-7 and 72 H-8. No gray-pink color differentiation was visible. A search was made of all samples under 40 for foraminifers and distoms, however none were found.

#### Conclusions

For the following reasons it is concluded that the sediments are lacustrine in origin.

 The textural variations in samples 72 M-7 and 72 M-8, and the sedimentary section indicate probable varying.

 The laminated bedding of gravels and silts in the meander scar describes non-marine depositional environments.

 The absence of forestinfferm and distons, while not necessarily proving a lacustrine environment, certainly is evidence for non-marine deposition.

The altitudinal differences in 72 M-5-6 and 72 M-7-8 might explain the higher proportions of clay in the former two samples. At the point of higher elevation, (Fig. 2-2) and thus later deposition, meluwater flow had probably slowed and sediment transported to the same locale would have been considerably finer. No other compositional variations were noted; this is logical since it is assumed that meltwater flowed over bedrock and outwash of the same derivation, that is, down the upper reaches of Indian Brook valley and Burnt Berry Pond valley. Since samples 72 M-7 and 72 M-8 were obtained from only the top few feet of an extremit mans of sediment, it may be that the

lower sections of the deposit are marine in origin. Based on the rather superficial analyses presented, it is obvious that the possibility should not be ruled out and that further work is needed to properly evaluate the problem.

### THE LOWER HALLS BAY DELTAS

#### Burnt Berry Brook and Dock Point

A raiged delts remnant is located 1.4 miles (2.3 kms.) mortheast of Burnt Berry Pood. Talands on the delts surface are 250 feet (75 meters) a.s.I., however the dissected scarp is nearer 180 feet (54 meters) in elevation. No field data were collected on this feature, but from an air photo analysis it appears that the delta was deposited by molt-water issuing from an ice from located in or near Burnt Berry Pond; formain around the delta remnant is humbocky and kettled, suggesting. stagmant ice.

Two miles (3.2 kms.) east of the above deposit there is a raised delta at Dock Point. Scarp and surface heights are similar to those of the Bornt Berry Brook delta. A 50 feet (15 meters) erosional terrace is visible man the bottom of the Dock Point scarp and continues along the southwest coast of Halls Bay.  $\sigma$ 

#### West Pond

A raised delta remnant, immediately due west of Riverhead Brook was the subject of field analysis (Fig. 2-6). Various elevations were taken on the main delta scarp; near the easternmost edge (Plate 10) a height of 200 feet (60 meters) was recorded. Several kettles are located on the surface of the feature; the bottom of the east-west





PLATE 9. BEDDED SILTS AND CLAYS NORTH OF INDIAN BROOK BRIDGE, 93 FEET (27.9 METERS) A.S.L.

PLATE 10. THE WEST POND TERRACE SHOWING 210 FDBT (60.3 METERS) AND 53 FDBT (15.9 METERS) LEVELS. RIVERHEAD BROOK IS INCISED IN THE LOWER LEVEL.



trending kettle nearest the T.C.H. is 109 feet (32.7 meters) a.s.l. The slopes of the kettle are rounded and show indications of fluvial or marine action.

North of the highway and southwest of Eaten Point a 178 feat (53.4 meters) level was recorded. The top of the northern section of the delta is hummocky and was measured at various points as 200-210 feet (60-60.3 meters). The base of the main delta scarp, southwest of Eaton Point is 53 feet (15.9 meters) a.e.1.

Hajor cuts in the delta are thickly covered with acress making etrike and dip measurements difficult to obtain, however one orientation of 345°, 35° R was recorded on the main West Fond terrace at an elevation of 100 feet (30 meters). This coincides with a strike and dip of 345°, 32° R obtained in the 50 feat (15 meters) terrace at the mouth of Riverhead Brook. Foreast gravels exposed in the 50 feat (15 meters) terrace are loosely compacted with sub-rounded to rounded publies 3-6 inches (1.3-15.3 cms.) in diameter. On top of the 50 feet (15 meters) terrace are g feet (2.7 meters) of postglacial lag gravels that contain younded stomes 1-12 inches (2.5-30 cms.) in diameter;

#### Sugarloaf

A traverse was completed from west to east across the Sugarland terrace (Fig: 2-6, Flate 11) and elevations have been recorded from several locations. No bedding planes were visible, however. At the break in slope on the west wide of the terrace a height of 58 feet (17.4 meters) was noted while on the eastern and, towards South Brook, a break in slope occurs at 49 feet (14.7 meters). Outcrops are vinthle

on the delta surface at 309 feet (92.7 meters). The main delta scarp is 212 feet (63.6 meters) a.s.l. while further back on the terrace a recording of 245 feet (73.5 meters) was made. Modern stream dissolution of the delta scarp is extensive, but despite this, the aurface is quite even and no major kettle depressions were encountered either during the traverse of from air photo analysis.

# South Brook

Directly east of the town, the main acarp of the South Brook tarrace is 249 feet (74.7 meters) a.s.l. Further south near the T.C.H., the height of the delta drops to 218 feet (65.4 meters). Two erostonal terraces are located within the town. The upper is 38 feet (17.4 meters) a.s.l. at the main break in slope and the lower was measured as 29 feet (8.7 meters) at the town wharf. Only one cut was visible on the main scarp (Plate 12). The six feet (1.8 meters) of topsets or lag gravels which cap the terrace at this point are well rounded and contain stones up to 1.5 feet (.45 meters) in diameter. One orientation of 020 dipping 35° L was obtained,however, it is thought that this conforms to slope bedding rather than true dairs forcests.

In the South Brook and Riverhead Brook velley bottoms, bedded glaciofluvial material was noted. No specific orientations were obtained and the texture of the deposits altered with location. In certain cuts, the glaciofluvial material was capped by varying thicknesses of postglacial lag gravels.

# White Point

The top of the delta at White Point (Fig. 2-6) is 220 feet



PLATE 11. THE SUGARLOAF TERRACE AND SUGARLOAF. PHOTOGRAPHED FACING NORTH. TERRACE HEIGHT IS 250 FEET (75 METERS) AT THE LEFT.



PLATE 12. THE SOUTH BROOK TERRACE WITH BEDDING ORIENTED 020°, 35° L.

(66 meters) a.s.1. A break in slope occurs at 110 fest (33 meters) with an erosional scarp at 54 feet (16.2 meters). A third level was noted at an elevation of 20 feet (6 meters). No bedding was visible on the face of the White Point delta, however a newly opened borrow pit on the highway midway across the terrace has good exposures of topsets and foresets. Orientations of 330-345 were obtained with dips ranging from  $7^{9}$ -13° R. It is not possible to be more specific in describing sediment texture in this pit since material varied from sandy cross-bedding to foresets with stones up to 10 inches (25.4 cms.) in dimeter.

#### Shell Samples

In early summer 1972, D. R. Grant recovered a sample of *Exiconus* which was subsequently dated at 12,000 ±220 B.P. (G.S.C. - 1733). The shells were located 55 feet (17,5 maters) s.s.1. in a stony pelite located at the front of the Sugarloaf terrace.

Two shell samples were collected in the Halls Bay areą by the writer. Sample 72-155, Mytičius sdulia, was located in a sand leps in foreset gravels on Kiverhead Brook (Fig. 2-6) 29 feet (8.7 meters) a.s.1. Sample 72-755 contained perfect specimens of Hiatelia avotica (Fiate 13). The sample was removed from a cut in story pelite 15 feet (4.5 meters) a.s.1., in Spring Gove at the eastern end of Halls Bay (Fig. 2-6). Pieces of the shells were visible on the surface of the cut but most were dug from the compact silt. Both these samples are estimated to be similar in age to G.S.C. - 1733 and are significant in that they date the innermost series of glaciomarine deltas of the morth cast comet of the idiand.

Henderson (1963) obtained three shell samples north of the field area which were subsequently dated. Sample G.S.C. - 55 (11,520 ± 180 B.P.), Macoun calcarwa, Mga transista and Higtella arctica, was obtained from a silty clay 160 feet (48 meters) a.s.1. in the Baie Verte River. G.S.C. - 75, collected from till-like material 35-42 feet (10.5-12.6 meters) a.s.1. in Middle Arm, Green Bay, was dated at 11,950 ± 170 B.P. At Southwest Arm, Green Bay, a sample of Higtella arctica and Mga transata was deposited 11,880 ± 170 B.P. The shells were located in silty bottoment clays 40 feet (12 meters) a.s.1.

# Conclusions and Discussion

The shell dates from Green Bay and Halls Bay are roughly equivalent. No one date falls cutside the upper or lower statistical error of the other samples. Excluding sample G.S.C. - 55 all were leas than 60 feet (18 meters) a.s.l. and thus represent deep water growth in delta bottomsets. The dates indicate that Green Bay and Halls Bay were deglaciated within a fairly marrow interval of time approximately 12,000 B.P. and that the deltas were not, as has been suggested by Jenness (1960) and Lundyvist (1965), deposited in ice dammed lakes. The following sequence is proposed for the deglaciation of Halls Bay.

A calving the front rapidly deglacisted Halls Bay until it became land fast in Indian Brook and in the valleys to the south where it remained while delta deposition took place at Springdale and lower Halls Bay. Bottomsets of the deltas fronting on Halls Bay are equivalent in time and were deposited at ice recessional position 1 (Pigs. 2-2, 2-6). Construction of the Springdale delta occurred from

lateral meltwater in Indian Brook valley and deposition across the bedrock ridge northwest of the main terrace, (Fig. 2-1). Material was derived from the valley systems to the west and along the north side of Indian Brook valley.

The silt units throughout Indian Brook valley may represent a lacustrine environment during this retreat or later, though the possibility of a marine origin for the lower portions of the large mass of mediument near Indian Brook Bridge is not discounded. Marine waters cartainly menetrated some distance up the Indian Brook - Burnt Perry Brook valley for delta deposition to occur southwest of Springdale. Purther manlysis is necessary for complete clarification of this problem.

As ice represent continued, major marine deita deposition took place at Burnt Berry Brook, Dock Point and White Point with continued minor deposition at the head of Halls Bay (Position 2, Figs. 2-2, 2-6). Although there is little remaining evidence to confirm the idea, it is possible to speculate that at Position 2, a 250 feet (75 meters) terrace was constructed just southeast of the T.C.H. - Springdale highway intersection. This would have paralleled the existing terraces at Burnt Berry Brook, Dock Point and White Point.

Finally, meltwater from ice located in West Pond and South Pond valleys (Position 3, Fig. 2-6) was responsible for the larger proportion of West Pond, Sugarloaf and South Brock terrace construction. Greatest amounts of ice recession between positions 1 and 3 took place on interfluves with relatively slow retrest in the valleys. Kettle holes and hummocky moraine on the highlands sround the deposite indicate that

ice fronts were near the deltas during their construction.

It is also concluded that the delta system at the head of Halls Bay which includes the West Point, Sugarionf and South Brook terraces, completely filled the lower perfecter of the bay. Confinued glaciofiuvial and postglacial erosion removed large portions of the delta from the two valleys (Plate 14). British Admiralty Chart No. 4591 shows depth soundings of 30-40 fathoms (54-72 meters) in lower Halls Bay with a widden drop to 103-120 fathoms (185-216 meters) immediately north of Dock Point. This suggests major fjord infilling by the eroded deltaic material.

49

During and after the process of ice removal from the field area, isostatic rebound caused the terrain to be uplifted. From recorded elevations of the Halls Bay deltas the marine limit or limit of postglacial emergence relative to present sea level is 250 feet (75 meters). Various sources cite sea levels for 12,000 B.P. as being about 172-224 feet (52-68 meters) below present levels. Walcott (1972) suggests a level of -224 feet (-68 meters) based on Grant's (1970) work on the Great Northern Peninsula of Newfoundland. Flint (1971) indicates a sea level of -124 feet (-68 meters) based on Grant's (1970) quotes Godwin et al. (1958) as predicting general sea levels of -172 feet (-52 meters) at 12,000 B.P. All the above values are based on curves constructed from radiocarbon-dated shells and material assumed to have been existing at sea level. This is a major wakings in that certain species substated at considerable depths depending on the temperature and eshifty of the water.

For this reason custatic curves may overestimate the amount:



PLATE 13. SAMPLE 72-255, OF *HIATELLA ARCTICA*, LOCATED AT SPRING COVE 15 FEET (4.5 METERS) A.S.L.



PLATE 14. LOWER HALLS BAY SHOWING; SOUTH BROOK TERRACE, LEFT, SUGARLOAF TERRACE, CENTER, AND THE SUGARLOAF, LEFT CENTER. of sea level change by as much as 60 feet (18 meters). However from the approximate custatic data it may be that c. 440 feet (133 meters) of isostatic uplift has taken place since 12,000 B.P. of which 190 feet (57 meters) has been obscured by a custatic rise in sea level during the same postglacial interval.

Erosional terrates occur at 220, 200, 180, 50 and 30 feet (66, 60, 54, 15 and 9 maters). (Table 2-2). All except the 50 feet (15 meters) level are poorly defined and do not appear on all deposits in Halls May. The 50 feet (15 meters) terrate is continuous throughout the area and represents a wajor still-stand. This is not to suggest that the other still-stands were not important but that evidence for them is scanty, having been removed by later marine erosion or mass wasting of whoosolidated material.

DELTA SYSTEMS THE AISED SHORELINE FEATURES ( SLEVATION

South Brook	249' (74.7 m.) (U.M.L.	218' (65.4 m.)			58' (17.4 m.)	29' (8.7 m.)	
Sugarlouf	245' (73.5 m.) (U.M.L.)		212' (63.6 m.)		49' (14.7 m.)		
West Pond	250' (75 m.) (U.M.L.)		200' (60 m.)	178° (53.4 m.)	53' (15.9 n.)		
White Point	250' (75 m.) (U.M.L.)	220' (66 m.)			54' (16.2 m.)	20' (6 m.)	
Dock Point	250' (75 m.) (U.M.L.)		•		50' (15 m.)	•	
Burnt Berry Brook	250' (75 m.) (U.M.L.)	•		180' (54.3 m.)			
Springdale	252' (75.6 m.) (U.M.L.)	220'" (66 m.):	210' (63 m.)	180' (54.3 m.)	50' (15)	3. 	

.M.L. - approximate upper marine limit

TABLE 2-2

# CHAPTER III

#### ESKERS AND CREVASSE FILLINGS

# Introduction

The purpose of this chapter is to relate a zone of eachers and crewasse fillings (Fig. 3-1) to the overall pattern of deglaciation of the Halla kay - Topsails area. The central part of the field area is marked by a number of ridgen of apparent glaciafluvial origin. Air photo analysis reveals that while some of then appear to be crewasse fillings, as suggested by lundquist, (1963), others bear a greater similarity to true eakers.

Generally, all the deposits studied are on the High Central Plateau with the larger and more prominent ones being located at the northern boundary, specifically near Barney's Brook and Sheffield Hill. The larger ridges near the edge of the plateau are up to 3 miles (4.8 kms.) long while further south lengths are about 0.8 mile (1.3 kms.).

Lundqvist (1965) describes one of the features lying outside the present study area as "a low but sharp ridge of a type that in Sweden is considered to have been formed in open-crowsesses in the ice..., and is of limited value for the determination of ice flow" (p. 296). He observes that all "eskers" in the chrrent field area are of this origin. In order to interpret the High Central Plateau deposits adequately, it is necessary to describe various possible methods of formation.



Eskers

While anglecial and supraglacial processes have been suggested, Fint (1971) favors a subglacial origin for the deposition of esker material, whereby, is a late phase of degladation, surface seltwater seeps through the ice seeking the lowest possible channelways. Water and sediment transported via this route is deposited partly along the beds of such channels and partly into ponded bodies of water at the glader terminus.

Levis (1949) suggests that externally ponded water is not necessary for frontal or near-frontal eaker deposition. He observed an eaker in the process of formation at a location where a contemporary valley glacter terminates on a gentle slope covered with solifluction debries. Since the valley had not contained water recently, Lexis explains that ponding necessary for eaker development had occurred under the glacier and deposition had taken place in stagmant water dammed in ice caves.

Embleton and King (1970) envisege a similar method of opkor construction. Instead of requiring ponded water near the glacter terminus, they maintain that a reduction of terrain gradient under the ice or blocking of stream channels by the collapse of decaying ice may be instrumental in initiating the deposition of material in aubglacial channels.

A variant of the frontal and normal eskers mentioned above in the beaded esker which is formed by the annual addition of small mounds of sediments as an ice front retreats. The summer period of rapid=WHTer flow adds to each bead, while winter is represented by the
#### intervening gap.

### Crevasse Fillings

Doorhamp, and King (1971) describe a process of crevased filling on west Maffin Island where they observe that the irregularity of the deposits, in plan and profile, indicates that they are not formed by flowing water but that they represent the filling of cracks in dead ice by material being accumulated from the surface above. The slow wahing down into the hollows would account for diminution of ridge height towards the distal and of the ridge. From several publications (esp. Filmt, 1928) the following summary of variations in seker md crewase filling morphology is tendered (Table 3-1). The "table is self-explanatory and will not be elaborated on at this point."

### Morphology

#### Barney's Brook

The Barney's Brook "esker" is located 4 miles (6.4 kms.) northesst of Foretopsnil (Fig. 3-1) and is approximately 2.9 miles (4.6 kms.) in length. The feature trends 130 on its main section while mean what appear to be distal distributories the orientation is 070-080° (Plate 15).

On the section of the deposit east of Barney's Book, parched erratics are numerous. The granite boulders are 3-5 feet (0.9-1.5meters) in diameter and often have numerous smaller erratics stacked atop them (Plate 16). The crest of the feature is sharp and hummocky. Flope angles taken at sample points varied from a minimum of 16° to a maximum of 22°. Towards the ice-marginal end, near sample point 72 k-1, the deposit is 28 feat (8.4 meters) above the general terratn.phile at

	TABLE 3-1	**
DISTINCTIVE	FEATURES OF ESKERS AND CREVASSE	FILLINGS

#### ESKERS CREVASSE FILLINGS -Situated within and tributary to terminal or Closely associated with recessional moraines 1. 11. recessional moraines. and pitted outwash; lie beyond the former and incorporated within the latter. 2. Commonly several miles in length; height of Most are short, less than 0.25 miles (0.4 kms.) 2. 12-45 feet (3.6-13.5 meters). and range from 12-15 feet (3.6-4.5 meters) in height. 3. Usually (not always) trend parallel to the Trend in any direction regardless of ice 3. direction of ice movement and may be direction and are continuous rather than discontinuous. broken . 4. Commonly sinuous in plan and arranged in 4. Are not arranged in distributary systems but distributary-tributary systems. where associated with lake deposits marginal. to ice edge may merge with them: are generally straight though may bend sharply around kettle rime. 5. Trend uphill and cross divides with no change 5. Tops are horizontal and the features do not in the volume of the deposit. Crests are pass over divides. knobby and hummocky or undulating, rarely level for long distances. . 6. Creat elevations not related to surrounding Crest elevations are similar to surrounding deposits. . kame and kettle rims. 7. Material coarse, lacks sands and clays; Composition is similar to that of eskers: predominance of gravels and cobbles. side slope angles also similar. Visible bedding in transverse sections parallels slope at angle of rest, 18-30 degrees.

#### ABLE 3-1 (CONTINUED

ESKERS	CREVASSE FILLINGS
<ol> <li>Bedding variable with semistratified to non- stratified lenticular masses. Sections of till may cover seker-ascribed to collapse of tunnel roofs and melt of debris-laden ico.</li> </ol>	<ol> <li>Bedding is moderately foreset to horizontal. Sections never show costings of till.</li> </ol>
<ol> <li>Ridge crests trend upwards in the direction of former ice flow to reach a maximum at the distal and.</li> </ol>	9. Gradient slopes down in the direction of water flow indicating flow accumulation, i.e., ridge heights diminish towards the distal end of the deposit.



PLATE 15. THE BARNEY'S BROOK ESKER, PHOTOGRAPHED FACING SOUTH. FORETOPSAIL IS SHOWN ON RIGHT CENTRAL HORIZON.



PLATE 16. PERCHED ERRATICS ON THE BARNEY'S BROOK ESKERS, LICHENS COVER THE UPPER SURFACES. sample point 72 E-3, about 0.9 mile (1.3 kms.) further west, height decreases to 16 feet (4.8 meters).

The crests are devoid of vegetation with heathland and scrub covering the Sides and surrounding low areas. In general the terrain slopes gently to the north east and is bouldery in nature. Well weathered, pale red, equigranular "Topsalis granite" (Neale and Nash, 1963) bedrock is visible and erratics of the same lithology litter the area. Near Barney's Brook "caker" the terrain becomes more hummocky, indicative of sblation moraine (Glacial Map, Chapter 5), though towards the northeest end of the feature the hummocks have a transverge linearity suggesting bighty dissected recessional moraines.

Although a more detailed textural analysis will be presented . lates in this chapter, it was observed that surface material of the deposit appeared well-sorted with a distinct lack of clasts over 1-2 laches (2.5-5 cms.) in disseter. The possibility exists that sorting was not a result of glaciofiuvial deposition but rather of postglacial op periglacial action, however since sediment size did not increase substantially with depth, the initial essemption is probably correct.

Sheffield Hill

A prominent gladiofluviel ridge located 0.9 mile (1.4 kms.) southeast of Sheffield Hill, is morphologically similar to the deposit described at Barney's Brook. It is 1.1 miles (1.8 kms.) in length, oriented 355, and increased in breath and height towards the northnorthwest (Plate 17). of rounded to sub-angular granite and granodiorife 1-2 inches (2.5-5 cms.) In diameter. Similarly, from an initial inspection, there seemed to be a lack of argillaceous material or large boulders within the deposit, though as was noted at Barney's Brook, erratics litter the surface. In places several smaller ertatics are perched of the larger base boulders.

At the low southern and of the "eaker" heights wary from 17-18 feet (5.1-5.4 meters). Near the center of the feature height increases to 23 feet (6.9 meters) and alone angles are 25° (west) and 26° (east). The northernmost section has steep/Sides and a sharp creat about 10 feet (3 meters) wide while height was leveled as 41 feet (12.3 meters). The surface of the deposit is david of vegetation at its southern extremity but is covered with scrub spruce and jumiper invards Sheffield Hill. Valley axial gradient is poorly defined at this location. Minor valley slopes are oriented northeast and southwest, though agior topographic slope is to the northwest.

#### Foretopsail

0.6 mile (0.96 kms.) west of Foretopsail there is a smaller eaker of crevases filling remnant 0.5 miles (0.80 kms.) in length. (Plate 18). It is origined 335, has a low, broad crest and side slope angles of 19° (south) and 18° (north). Although the feature is 11 feet (3.3 metras) above the immediate terrain, surrounding ablation morales is generally of the same, or greater, height. There are no surface boulders and vegetation is minumal. A pit was dug to a depth of 2.5 feet (0.75 metors) and no bouldery metrail was encountered, rather the



PLATE 17. THE SHEFFIELD HILL ESKER, PHOTOGRAPHED LOOKING NORTH.



PLATE 18. THE FORETOPSAIL AREA SHOWING THE CREVASSE FILLING REMNANT (LEFT) AND BOULDERY TERRAIN OF THE HIGH CENTRAL PLATEAU. sediment was well sorted and sub-rounded containing clasts from 0.80-1.3 inches (2-3 cms.) in diameter.

#### Air Photo Deduced Morphologies

A ridge fituated in the valley north of Little Sandy Pond appears from "bir photo analysis to be morphologically similar to the Barney's Brook "esker," 1.5 miles (2.4 kms.) to the west. The deposit follows the valley axis at 340° and is dissected by meltvater channels and existing streams (Fig. 3-1). Its overall length (1.2 miles, 1.9 kms.) and orientation further indicate a relation to the Barney's Brook complex. The Little Sandy Pond valley is filled with glaciofluvial ... meterial and surrounding drift is abundant.

Two miles (3.2 kms.) east-southeast of Foretopsail there is a deposit which may be a true crevasse filling as described by Lundqvist (1955). It is approximately 0.8 miles (0.98 kms.) in length and branches at various points. Part of the feature trends 355° while other sections are oriented 010° and 035°. The ridges are highly dissected and pitted with kettle holes.

Frominent meltwater channels have been plotted from the air photos (Figure 3-1). Near Sheffield Hill the pattern indicates a northerly flow of meltwater, while in the Barney's Brook area flow was radial from the surrounding summits with final flow to the northnortheast down Barney's Brook and West Brook.

#### Textural Analysis

Samples were collected from the deposits at Barney's Brook and Sheffield Hill. All were taken unsieved from a depth of 2.5-3 feet (0.75-0.9 meters) on the slopes of the features. Four samples were analyzed; three from Barney's Brook "esker," 72 E-1 (most easterly), 72 E-2, 72 E-3, and one, 72 E-4; from the Sheffield Hill "esker." (Fig. 3-1)

A standard sizve analysis was completed at whole 4 intervals. as outlined in Appendix 1. Individual percentages were calculated for the sample weights and the results were plotted on Cumulative (Arithmetic Ordinate) Curves (Fig. 3-2) and Prequency Curves (Fig. 3-3). » A pipette analysis was not completed on the fractions-less than 40 since only a general indication of (or lack of) glacifilwill action was required as a result, which would be obvious from the standard sieve analysis. A fabric analysis was estempted at sample point 72 E-4 (Plate 19) but the sediment was neither well consolidated norwere the publics clongated enough to complete an analysis with any degree of accuracy.

Results

The following results were obtained from the textural analysis (Table 3-2).





#### TABLE 3-2

Sample	Inclusive Graphic Standard Deviation	Mean	Kurtosis	Inclusive Graphic Skewness
72 E-1	1.24¢ Poorly Sorted	+1.07¢	0.95 Mesokurtic	0.06 Near Symmetrical
72 E-2.	1.96¢ " "	-0.13¢	1.21. Leptokurtic	. D.23 Fine Skewed
72 <sup>°</sup> E <del>¬</del> 3	1.530 " "	<b>+0.97</b> ¢	1.11 Leptokurtic	0.25 " "
72' E-4	1.09¢ " "	-1.58¢	0.91 Mesokurtic	0.30 " "

#### CALCULATED DATA FOR HIGH CENTRAL PLATEAU SAMPLES 72 E-1 - 72 E-4

All four samples have primary modes occurring in a zone of very coarse sand, with an abrupt break in slope on both sides of the modal peak (Fig. 3-3). It is possible that the primary modes may have been influenced by the upper limit placed on the slewing, though in three of the four samples analyzed a distinct absence of material greater than -20 occurs. Motably, though, there is a lack of fine sand (2-30) with a slight increase in very fine sand (2-42) in all samples (fig. 3-3).

From the Cumulative Curves (Fig. 3-2), one may observe that at least 90% of the sediment is confined to material coarser than 2.60, that is, with a sharp break in slope of the curves within the send categories (Table 3-3).



PLATE 19. SAMPLE FIT 72 E-4 SHOWING SUB-ROUNDED, GLACIOFLUVIAL SEDIMENT IN THE SHEFFIELD HILL ESKER.

## TABLE 3-3 TEXTURAL BOUNDARY (\$ UNITS) AT 90% CUMULATIVE WEIGHT

SAMPLE	· · ·	¢ UNITS
72 E-1		+0.60¢
72 E-2		+2.600
72 E-3		+1.200
72 E-4	11 A.	0.000

Since the samples were not obtained from contiguous bedding planes, it would be unwise to intimate any spatial relations for deposition, however the lack of very fine sand and silt is a nutable general feature. Samples 72 5-2 dfm 72 8-3 may be described as lebrokurtic.

Possibly this is a result of the large amount of material in the -20 -60 range. Samples 72 E-1 and 72 E-4 are mesokurtic and thus have more even sediment distributions. All except 72 E-1 are fine seawed. This does, not imply that they contain a large amount of silt of very fine sand but rather, indicates a skewing of the frequency curve to the fine section; however all samples are in a closely defined range and are comparable in the general scope of positive or fine skewness.

Stereo Microscope Analysis

From an examination of the 2¢ sections of the samples under a stareo dicroscope it was noted that samples 72 E-1, 2 and 3 had roundness values of 0.30 (Powers (1953) scale); that is, they were sub-angular with a high degree of sphericity. Sample 72 E-4 was more angular (0.20) with a high sphericity. A tarnish, probably due to postglacial chemical weathering, was present on most grains though a slight polish was visible on the quartz fraction, the latter being indicative of abresion by water transport.

As a general result of the analyses there would seen to be a certain amount of sorting by water and a lack of fines and very coarse material in the samples.

#### Conclusions and Discussion

From the textural analysis, morphologiss, and literature, it is knoncluded that the two features at Barney's Brook, which includes the deposit in Little Sandy Fond valley, and the Sheffield Hill formation ate true eskers. The reasons for deciding so are:

 The high number of undisturbed perchadgerratics on the deposits at Skeffield Hill and BarnBy's Brook. If the material had been deposited in open cravases it is unlikely that boulders would have lodged as shown in Plate 16. The erratics were probably deposited from englacial and supreglacial debris as ice stagnated in size.

2. The well washed granular material and lack of a very fine sand-silt fraction would also indicate esker-like composition. These signify flow deposition and a removal of specific sediment fractions rather than infilling and noor sorting of material.

 The dimensions of the features are greater than those outlined by Flint (1928) for crevesse fillings (Point 2, Table 3-1).

4. In each of the locations the area surrounding the eskars is free of large amounts of solation mornine and kettles, the presence of which would imply crewasse fillings (Point 6, Table 3-1).

5. While no bedding was discovered, this does not preclude

classification of the deposits as eskers since Flint (1928) notes that they may be semi-straified to non-stratified (Point 8, Table 3-1).

6. The association of the Barney's Brook - Little Sandy Yond valley complex with an area of recessional moraine conforms with point 1 (Table 3-1) which defines eaker location as being within, and tributary to, terminal or recessional moraine. This is possibly the weakest of the criteria listed for deciding the feature origins. While the point is valid for smaller localized valley eakers, deposits the size of Munro eaker (lat. 48° N, long. 80° V) which is 250 miles (400 kms.) long and 1-6 miles (1.6-6.4 kms.) wide (Lee, 1965) would obviously be associated with multi-variate glacial deposits.

The orientation of the eskers seems unusual, since in an overall field context ice movement was to the northeast, however it may be explained with relation to more localized flow. Following an initial singe of coastal delta construction, ice retreated steadily until a pause occurred at the 800 feet 43 meters) contour delineating the High Central Plateau (Chapter 1 and Fig: 3-1). South and southwest of this line ice remained long enough for esker formation.

Neale and Mash (1963) may west-east stose and lee forms near the Barney's Brook source area indicating late east and west ice flow. Within thig zone, however, the terrain is flat. Any strong hydrological gradients would have been confined to major valleys. Nee at this stage of deglecistion was thin and major hills presented parriers to the movement'of subglacial waters.

From the criteria outlined by Doornkamp and King (1971), thickening of an eaker towards its ice marginal or distal end, (Point 9, )

Table 3-1), it appears that for the Barney's Brook esker hydrological gradient was to the southwest and then northeast at its terminus. Major topographic obstacles were avoided and final flow was confined to the Barney's Brook Valley. Flow in the Little Sandy Popd - Barney's Brook esker was to the northwest. Orientation of the deposit conforms with the long-valley axis; roughly north with a northwest component at the distal end. Meltwater flow in the Sheffield Hill esker was to the north, a shallow valley context and with a localized hydrological gradieco.

These orientations are logical since ice was thicker several miles southwest of the eskers and in the case of the Sheffeld Hill, vicinity there had been a westerly component to ice flow at a late ? stage of deglaciation (Chapter IV). Further cast at Barney's Erook topographic gradients and valleys would have been pronounced enough to control late ice flow in an approximate northerly direction.

The direction of meltvater channels on the High Central Plateau (Fig. 3-1 and Clackal Map, Chapter V) cospare favourably with the orientations of the sekers. The channel patients indicate topographic flow (%s) to the northwest near Sheffield Hill, and (b) into the Barney's Brook drainage system further east. Thus the zone of eskers in indicative of late topographically controlled ice flow from the High Central Plateau.

Other features analyzed in the vicinity of Foreiopsail are crevesse fillings similar to those described by Lundqvist (1965). They are highly dissected, associated with kettles and ablation morates, tryfil at various angles within a specific deposit and are less than 1 wilk (1.6 kms.) in length.

# CHAPTER IV

THE KITTY'S BROOK - CHAIN LAKES MORAINES

#### Introduction

The Kitty's Brook - Chain Lakes valley system contains a series of large southeast-northwest oriented ridges. The field area extends towards the southwest to include these features. Although dominant ice flow considered in the total study was to the uortheast, it is from the Kitty's Brook - Chain Lakes area of the High Central Plateau that late topographic flow was directed downvalley to the southwest, and then northwest into Sandy lake (Plate 20). Thus, analyzing the Kitty's Brook - Chain Lakes features may be considered an important step in synthesizing the deglacial sequence of the field area.

Several early researchers made note of the general area. Coleman (1926) decided that the High Central Flateau had not been glaciated during the Wisconsin:

The general tablaind does not suggest placiation since if is a plain covered with prach bodders (SEC) of granits. and general just like-the underlying bedrock which shows as low bare ridges in places but is usually hidden by coarse and fing granitic derivs which seems to have originated in place. . . The greatly weathered surface of the mountains (Copasils) and the tablaind . . suggest that glaciation was probably of Kansan or Jerseyan age. (on. 205-206)

" Coleman's mistake in dating the Topsails area glaciation was in assuming that if the area had been covered by Wisconsin ide



PLATE 20.

THE KITTY'S BROOK - CHAIN LAKES AREA.

N.A.P. UNCORRECTED MOSAIC NO. RE-5992

APPROXIMATE SCALE 1 INCH = 4 MILES 1 CM. = 5.6 KMS.

the degree of periglacial and postglacial weathering of the "Topsails granite" would have been far less than it in fact is.

MacClintock and Twenhofel (1940) make specific mention of the Kitty's Brook area:

From Kitty's Brook Station to the top of the Plateau at Gaff Tonsail [there are] morainal hills 50 pm 200 feet (15-60 meters) high composed of gravel and till. Interspersed are kettle holes of equivalent dimensions, many of which contain lakes.

The Kitty's Brook moraines were built when tongues of the upland ice, cap projected part way down the valley. This stand . . . was either a halt in recession or possibly a re-advance. (p. 1745)

Prest et al. (1967) have mapped the ridges as ribbed moraine. large scale features with a transverse linear pattern giving a ribbed effect to the land surface

oroceas.

The individual ribs consist of bouldery ridges up to a mile (1.6 kms.) or more in length; 30-90 feet (9-12 meters) high with crests 300-1000 feet (90-300 meters) apart (Prest, 1968, p. 6).

Lee (1962) attributes the formation of ribbed moraine to an overriding of basal till. The woulding-process is described as occurring during re-advance when the ice glipped over unfrozen till with quantities of the material being built up or undercut at the ice edge by a bulldozer effect. As the material became more compact and resistant, ice moved up and over the ridge, continuing its advance by repeating the

Ribbed moraine has been associated with drumlins and fluted ground moraine; the latter may be linked to an increase in altitude and a change in certain factors of formation, such as flow from high to low pressure areas (Cowain, 1968)-

The uncertain origin of the features and the importance of the zone in the sequence of deglaciation of the field area mecassitated a traverse in the Kitty's Brook - Chain Lakes area. Several samples were collected for later textural analysis and a series of till febric diagrams were plotted.

#### Morphology

The gouthwest Chain Lakes area deposits are from 700-1500 feet (210-450 meters) wide in a north-south direction, span 1-1.3 diles (1.6-2.1 kms.) across the valley and are spaced 2000-2500 feet apart (Fig. 4-1, Plate 21). In the northeast, towards.Sheffield Hill, spacing between the features increases to 3500-5000 feet (1.1-1.4 kms.) while central lake Grainer ridges are discontinuous. Heights of the lineaments in the southwest area range from 30-90 feet (15-27 meters); at the northeastern extremity of the lake-chain heights increase to 130-140 feet (39-42 meters). The major northeasterly ironding trough in Which the features area located is 1.0-1.4 miles (1.6-2.1 kms.) wide and 150-200 feet (45-60 meters) deep. Many of the lineament creats coificide in elevation with the High Central Plateau at 1250 feet (375 meters) to the cast and with a nerrow planar remain west of the valley.

Towards the southwest, the slopes of the deposits are very smooth with slope angles approximately equal; (Plate 22) at the northeastern and of the valley system slopes are steeper, and concave in plan and profile towards the southwest, suggesting ice contact features. Lakes in the center of the valley are elongate while at eithe and of the system they are rounded.





PLATE 21. THE CHAIN LAKES RECESSIONAL-ABLATION MORAINES PHOTOGRAPHED FACING SOUTHWEST.



PLATE 22. CHAIN LAKES MORAINE 72-3. NOTE THE LOBATE CREST AND SHALLOW SLOPE ANGLE.

Since the deposits are concave up slope at both ends of the valley system orientations of 285-350 represent only a trend normal to the structural trough. Moraines on the east side of the plateau (middle-cast, Fig. 4-1) are oriented in a northeast-southwest direction; their heights and slopes are similar to those previously described.

### Textural Analysis,

The purpose of the textural analysis was to accurate if the deposits are glacial or glaciafluy(al, that is, whether the material is a glacial till shaped by a finfil ice thrust or outwash deposited by maltwater around stagaant ice bodies. Normally the latter is stratified. No stratification was visible in the deposite, therefore, in order to confirm the materials' origin, two samples were collected from cuts near the southwest end of Chain Lakes (Fig. 4-1). It was felt that these would be representative of sediment contained in the transverse deposite.

A pet sieve analysis was completed on the samples at whole \$ intervals and the results were plotted on Arithmetic Ordinate Cumulative Curves (Fig. 4-2) and Frequency Curves (Fig. 4-3). Colculations of the mean, inclusive graphic standard deviation, kurtosis and inclusive graphic skewness were completed as outlined in Appendix 1. The results are presented in Table 4-1.



1 m 1 4 4 1



#### lesuits

#### TABLE 4-1

CALCULATED DATA FOR CHAIN LAKES SAMPLES 72-2 - 72-3

Sample	Mean	Inclusive Graphic Standard Deviation	Kurtosis	Inclusive Graphic Skewness
72-2'	1.136	2.54¢ Very Poorly Sorted	0.80 Platykurtic	0.04 Near Sym- metrical
72-3	1.100	2.52\$"""	0,66 Very Platykurtic	0.08 "

The above results are indigative of a glacial till. The poor sorting and symmetrical Frequency Curves suggest the material has not been exposed to significant glaciofluvial action. The kurtosis values of .80 and .66 describe the bimodal quality of the sediments (Fig. 4-3). Both samples have a lack of material in the 3-40 range. This conforms with the sarameters for a glacial till described by German (1964), which show-a flattening of the Cumulative percentage Curves at 5-30 for various European till samples. While the 4 range used in the Cumulative Curves (Fig. 4-2) in this report is not as braid as that used by German, a similar 3-40 trough is visible on the Frequency Curve (Fig. 4-3).

#### Stereo Microscope Analysis

From a stareo microscope gnalysis of the 20 segments of the till samples, it was ascertained that the grains are angular to very

angular (.17, Powers (1953)) with a low degree of sphericity. There is little polish to the quartz content of both samples. This is indicative of a till ond of glacial abrasion which results in tiny angular irregularities or fresh fracture surfaces (Wolk, 1968).

#### Till Fabrics

To further test the various theories of the origin of the Kitty's Brook - Chain Lakes features, several till fabric analyses were completed on selected deposits (Fig. 4-4, a,b,c,d). If the features were ribbed moraine or recessional features, one would expect a primary mode normal to ridge creats since both features are . derivatives of ice flow.

Cowar (1968) describes the orientation of straight-ridged ribbed moraine clusts as "appearing in all cases to fall close to the direction of a glaciation" (p. 1150). For curved ridges he found that fabrics indicated movement related to "both the direction of glaciation and to some attribute of the ridge itself" (p. 1153).

Recessional mornine is a resultant dump of ground moraine by stagnating ice but since it has been carried along during advance it would show some clast orientation paralleling final flow.

Bailway cuts were used as sites where possible, and where not, fabric pits were dug to a depth of 36-48 inches (90-120 cms.) in an attempt to avoid zones affected by frost heaving and weathering. Feature trends were noted in the field while primary and secondary modes were visually interpreted from Rose diagrams of the data, (Fig. 4-5, a,b,c,d). Means and standard deviations were obtained from









the till fabric diagrams and application of the relevant calculation outlined in Appendix 1. The results are presented in Table 4 - 2.

Results

#### TABLE 4-2

Sample	Mean	Standard Deviation	1° Mode	2° Mode .	Feature Trend
72-1F	010	30°	350	090	. 010
72-2F	010	42°	355	055	010
72–3F	005.	27° .	020	160	345
72-4F	325	22°	330	200	320

TILL FABRIC DATA, CALCULATED AND INTERPRETED ....

Primary orientations for 72-1F and 72-4F are approximately parallel to feature elongation. Fabric 72-2F has a weakly developed primary mode with a strong secondary mode normal to the ridge creat.

Frimary mode orientation for 72-39 is 180° from the assumed ice flow direction, that is it shows a marked plunge towards 010. This probably occurred because the till fabric diagram was constructed from clasts on the mortherly or up-valley slope of the ridge; all other samples were obtained from central positions or down-valley slopes. Otherwise the till fabric orientation at 72-39 has a primary mode sub-parallel to fee flow and a secondary mode parallel to feature clongation. In all cases two fabric modes have been observed; one parallel and one transverse to assumed ice flow from Kity's Brook valley. If one is to explain the features either as fibbed moratine or recessional moraine as previously discussed, the bimodal aspect of the till fabric disgrams need clarification.

Discussion

Glam, Donner and West (1957) note that in fill fabrics where there is a secondary mode, it is usually at right angles to the primary mode and may exceed it in hagnitude. In this instance the authors are assuming that a primary mode is parallel to fee flow rather than being the dominant orientation. They suggest that there are significant correlations between prolateness of clasts and a tendency to transverse modes, which they account for by collisions between more elongate stones.

Harris (1963) explains the relation of secondary modes to roughness of terrain. In areas of high relative reliaf he found that till fabrics had a high M.S.O.C. (Appendix 1) and poor model orientation. In the direction of ice flow. With a relative relief of 100-200 feet (30-60 moters) a notable development of transverse modes was observed. Similarly, Andrews and Smith (1970) describe the occurrence of a dominant mode at right angles to ice flow. They interpret the phenomenon with the idea that as ice amendeas a bedrock slope, clast rolling and thrusting becomes the main orientation-producing process; that is, a transverse mode is produced by changes in local flow conditions.

Rutter (1964) studied clast orientations in surging and normal glacier moraine. He found that normal glaciar deposits have strong preferred orientations in the approximate direction of ice flow.

In surging glaciers stone orientations are weak, and, where they are apparent, are not necessarily parallel to flow. One fabric site in normal glacier moraine whowed a definite transverse orientation. This he explains by observing that the sample site was on a steep slope and may have suffered post-depositional slump or alternately, consist of ablation debris deposited at the ice margin. While his ideas are worthy of note, it would be unvise to place too much explasis on them since the Xitty's Brook - Chain Lakes deposits are unlike those if Rutter's area.

Drefmanis (MacClinteck and Dreimanis, 1964) describes a series of till fabrics with two maximum orientations separated by an angle of 60°. One represents the penultimate glacial movement, the other defines the latest, which re-oriented at least half of the elongate pebbles parallel to the fabric maximum of the overlying till.

It is falt that the process described by Dreimmis is not applicable to the Kitty's Brook - Chain Lakes area, since here orientations were obtained near the surface of a contiguous till unit. One might suggest that the transverse modes were obtained from penultimate ice flow over the ridge north of Chain Lakes towards white Bay, however, it is unlikely that such an orientation would survive later re-working in a near-marface till unit.

The following mechanisms may be responsible for the strong transverse secondary modes in the Kitty's Brook - Chain Lakes area deposits:

- Re-orientation of prolate clasts during final flow. Unfortunately this cannot be tested since no record of individual clast

shapes and their orientation was taken.

- Height above the valley floor. This is probably the most likely reason since all till fabric sites were 100-150 feet (30-45 meters) above the valley floor and relatively near the bedrock walls. No pits were dug on the lower central remnants of the features.

Andrews and King (1968) state the writer's view in the following:

A till fabric is a sample of a larger unsampled set of pebbles and since fabrics may vary markedly within the same till exposure, single sample, single site results should be used with caution. (p. 458)

Obviously many more till fabrics analyses of the Kitty's Brook - Chain Lakes ridges need to be completed before one can concludively interpret their dominant fabric orientations.

#### The Kitty's Brook Glaciofluvial Deposit

At mile 343 on the C.H.R., on the north side of Kitty's Brook valley there is a small outwash delta or fan (Plates'23 and 24). Coreset bedding in the deposit strikes 020, dips 16° Å and contains large stones 1-3 feet (0.3-0.9 gaters) in diameter. On the valley floor (Plate 25) and in the topset material, large well rounded to subrounded boulders 2-4 feet (0.6-1.2 meters) in diameter are visible. These are indicative of large volumes of water that must have poured down the valley late in deglaciation.

Air photo analysis of the festure indicates several moltwater channels that may have contributed to its formation. The deposit appears to be a delta formed in a scall glacial lake, dammed up between two of the morainal features (Fig. 4-1), and since amptied by incision of Kitty's Brook. The deposit is presently


PLATE 23. C.N.R. MILE 343, THE KITTY'S BROOK DELTA.



PLATE 24. VIEW OF THE KITTY'S BROOK DELTA PHOTOGRAPHED FACING NORTH. BEDS ARE ORIENTED 020°, 16° L .



PLATE 25. KITTY'S BROOK VALLEY, (NOTE LARGE WELL-ROUNDED BOULDERS). PHOTOGRAPHED FACING NORTHWEST. being used by Canadian National Railways as a borrow pit and it is impossible to determine the former extent of the delta, though since it is nearly 60 foot (18 meters) thick deposition probably continued for a considerable period of time.

# Conclusions

Several features of the ridges prohibit their classification as ribbed moraine. Clast orientation is not well mough defined to meet the parameters outlined by Cowan (1968); also their morphological dimensions and spacing are greater than described by Prest (1968). From their morphology and the textural analyses, the following system is proposed for their construction.

The High Central Plateau was an ice dispersal centro in the late Wisconsin. During deglaciation, ice flow may have been towards the north or northwest as evidenced by eakers and esitwater channels (Chapter III). Ice pouring off the plateau flowed morth-northeast towards Sheffind Lake and southwest down Kitty's brook valley (Plate 20). The large northeset-southwest oriented till ridges east of Chain lakes (Fig. 4-1) may be realized to topographic flow from the plateau into the valley system. The contact alopes in the lower motheast sections of Chain Lakes valley suggest that rather than forming a series of true receasional motences, ice stagnated in large blocks behind the deposits, giving them their elongated, concave-convex forms. Hence the deposits may be classified as regissional-ablation moraine.

14 . The delta in Kitty's Brook valley was deposited just before a

94

final glaciofluvial stage which connected the individual ponds and .

formed a complete drainage system.

# CHAPTER V GLACIAL MAP COMMENTARY

This chapter serves to explain the classification and symbols used on the glacial map of the field area (pocket, back cover). While the preceding chapters of the thesis deal with specific geomorphic problems, it is hoped that the map will present a more complete picture of glacial events. Both drift prospecting and geochemical sampling require such an interpretation as a starting point. Since the field area is part of the Central-Mineral Belt and heavily drift covered, the following explanations used in conjunction with the map may provide a useful tool for future mineral exploration. It is recommended that the map be used with reference to Figure 6-1 since the features are not all contemporaneous and ice flow direction changes with stages of deglaciation.

Initially, a genefic terrain classification devised by Fulton et al. (1911) and used by the Geological Survey of Canada was planned as a mapping code. Unfortunately the system has several disdivantages. It does not focus on glacial landforms and events but rather presents a consolidated view of the landscape within defined boundaries. Also since the classification consists of a genetic category, morphological modifier and textural modifier, it is not easy to interpret the system unless one is completely familiar with the code. For example, eK. is interpreted as a flurial environment where the acologic process

responsible for the category is still actively affecting the area.

The modifiers state that the zone is a sendy thermokarst plain. Consequently, a modified form of the Glacial Map of Canada (Prest, Grant and Rampton (1967)) and Fulton's classifications has been attempted.

Unlike either of the terrain classifications mentioned, morainal zones have not been bounded; rather gradational symbols have been used that connote densities or marginal zones of specific features. The following list presents some of the terms used with brief explanations; others are self explanatory.

#### Delta Surface

The delta surface describes the raised deposits in a modern context. No differentiation is made under this heading for terrace levels in evidence on the main scatp. Landward boundaries were constructed at bedrock outcrops or distinct breaks in slope.

# Erosional Terrace

A well defined terrace has been mapped at the 50 feet (15 meters) level. Other erosional features exist (Chapter II) but are not continuous on all delta deposits and for reasons of scale have not been included.

# Kettles

Kattle holes have only been indicated when they are located on or near delta surfaces. Numerous kattles occur in several areas, but again because of scales and densities of the features they have not been mapped.

97

#### Outwash

Included within this denotation are materials deposited proglacially and derived specifically from meltwater as ice retreated inland. Essentially these are cross-bedded sediments of varying grain sizes deposited in walley trains.

# Major Meander Scar

The symbol has been used at only one location and describes a feature that might be mistakenly interpreted as a raised delta remnant.

# Ribbed Moraine

Ribbed morains has been described with none dotail in Chapter IV. Within the field area it is confined to shallow valleys and associated with elongate lakes which visually amplify the ridges. Prest (1968) describes the moraine as occurring on the flamks of drumlinoid ridges and eakers or on the low ground between them. In west-central Newfoundiand ribbed moraine is generally associated with drumlinoid ridges.

# Drumlins and Drumlinoid Ridges

True drumlins with steep stoss end and tapered distal slope are thinly distributed in the field area. Features mapped as such may or may not be rock-cored but their external morphology is similar to the classical dome-shaped till structure.

Drumlinoid ridges are more numerous and include narrow elongate rock-cored ridges tapered at either end, though the stoss end may be more blunt than the lee (Prest, 1968). Where defined by a stoss and lee form, direction of flow has been inditated, however several of, the drumlinoid ridges are not well developed and are located where the

-98

sense of ice flow is not immediately certain.

Flint (1971) observes the following for drumlin forms and ice

flow:

 The presence of such forms establish the existence of an actively flowing glacier at the time of formation in contrast with ineccontact drift which indicates comparatively inactive ice.

2. The long axis of the streamline forms are a more reliable indication of the general direction of movement of a former glacier than are striations because they are less influenced by local topography (p. 106).

# Hummocky Disintegration Moraine

Lundqvist (1965) classifies such a deposit as an irregular hummocky ablation moraine. Freet (1968) refers to an original work by Gravenor and Kupsch (1959) when he describes the features as "high relief, irregularly mounded and deeply pitted type of non-oriented hummocky moraine which is believed to form as a result of glacier desintegration or stagnation" (p. 11). In the field area deposits are from 75-100 fest (23-30 meters) high though Prest reports extreme cases of 200 fest (60 meters) in other locations. The deposit near Gaff Topsail are not extensively pitted but rather are "equidimensional mounded surfaces," (Prest, 1968) on a plateau. The lowlands around the moraine are bogsy and contain numercus shallow lakes.

#### Ridged Ablation Moraine

Hore prevalent is a type of disintegration moraine described by Hoppe (1952). Areas of such deposits contain:

 ridges up to 100-115 feet (30-35 meters) in height. (In west-central Newfoundland ridge heights are in the range of 50-60 feet (15-18 meters). 2. 'dead ice hollows.

3. moraine plateaus.

In the above study the ridges are regarded as being formed inderneath the ice through the squeezing of water-scaked morainal material into basal cavities. The dead ice holfows indicate areas of residual ice, while the moraine plateaus are considered to be secondary features. Hoppe alternatively suggests the plateaus might represent original elevations of material below the ice. Ridged ablation moraine is confined to valleys and has little value for determining ice direction. (Hoppe (1952), Lundquist (1965)). Stalker (1960) notes that the moraine does not necessarily represent former ice sheet margins, but rather may denote changes in the rate of, or even halts in, the lowering of the murface of an ice sheet.

# Recessional Moraines

The category includes the recessional-ablation moraine described in Chapter IV and other marginal and lobate recessional feature remnats. The designation may include interlobate moraine or kase terrace remnants without differentiation, since many of the features were mapped without field analysis.

# Structural Lineations .

Structural lineations are not to be confused with drulinoid ridges. They indicate the orientation of outcrop ridges and bedrock trends. For the central section of the field area, elongation of the symbols indicates the limbs of major, synclinal troughs (Neale and Nash,

1963).

# Glacial Striae

Strise have been plotted near Halls Bay and include only those mapped by the writer. Others have been recorded by Neale and Nash (1963) and Lundqvist (1965).

Striations plotted on the map are:

1. Lower Wolf Cove, Springdale - 070°

2. Roche moutonnee south-southeast of South Pond - 035, 020°

3. East of Burnt Berry Pond - 090, 110, 115°

4. North of South Brook town - 020°.

5. Roberts Arm Road - 315, 330°

# CHAPTER VI

From the air photo interpretation and analyses applied to specific glacial features, if is possible to guggest a sequence of events for the late glacial and postglacial of west-central Newfoundland. The field area is more of a transact than an entity, thus some correlations are implied with surrounding areas, based on the research of others and currént field evidence for the "Balls Bay - Topasile" area. Ice fronts (Figure 6-1) are sproximations only and were plotted from ice-marginal deposits where possible and where not, were implied from the general terrain. The figure should not be considered as a spatially exact chronological, sequence, but rather as an indication of the manner in which ice retreated.

#### Glaciation

During the late Wisconsin maximum, ice flow in the field srea was to the northeast (Fig. 6-1) and north towards White May from an ice cap centered south of the field area on the High Central Platcau. The orientations of drumlinoid ridges are evidence of this flow, as are the zones of ribbed morains which are small, and confined to walley bottoms where late flow would have been strongest. Multi-directional strias indicate late flow which mice changed direction as it moved around during which the changed direction as it

It is considered that there was at no time an invasion of



Labrador ice either in the Halls Bay area or on the Burlington Peninsula to the northwest (See Introduction). Grant (1969) states that:

Laurentide ice from Labrador advanced southward over at least the louisand portion of the Northern Perinsual and perhaps 1000 fest (300 meters) up the flanks of the Long Range mountains as evidenced by unidirectional grooving, frominent rocks mountances, Labradorian erretics on the uplands near Roddickton and St. Anthony and a shelly drift that was spread videly over the area (p. 124).

Following Grant's analysis, if Lauentide ice had in any way affected the Burlington Peninsula, one would have to assume ice yrapping around the tip of the Northern Peninsula with a southerly component of flow as it crossed the Grey Islands (Pig. 5 - 2) and moved towards White Bay and east. There is no evidence for this hypothetical southeasterly ombiore flow in the Burlington Peninsula or Halls Bay areas.

Deglaciation -

Shortly before 12,000 B.P. ice retreated from Halls Bay, (as well as Green Bay), and within the field ares, a set of raised doltas was deposited at Springdale and lower Halls Bay. Kettles and ablation moraine indicate that ice fronts were near the deltas during deposition. This may be the assumed outer drift zone boundary outlined by Landqvist (1965). The writer faels however, that if a cosstal area is deglaciated quickly by calving ice fronts in the major bays and fjords with ice remaining just inland as in position 1 (Fig. 6 - 1), deltas would obviously demarcate the ice margin. This does not imply inner-outer drift zones but rather describes ice retreat sub-paralleling a land-sea margin. In the Halle Bay area a total of 440 (set (133 meters) of postglacial uplift has occurred since 12,000 B.P. and is represented by 250 feet (75ceters) net emergence after a custatic rise in the same period of 192 feet (58 meters).

During the delta-building stage, ice retreated from the coast by stagnating in the valleys and lowlands between positions 2 and 3 (Fig. 6. - 1). Ridged ablation moraine is found in the middle and upper sections of the valleys where ice was thick enough to cause basal squeezing necessary for its formation. In the area northeast of Sheffield Lake, ridged ablation moraine partially overlies an earlier zone of ribbed moraine. Radial estimater channels around the hills are further evidence of stagmant lowland ice and general thinning in early deglaciation.

At position 4 (Fig. 6 - 1)/a still stand is indicated. The zone is delimited by a group of eskers just inside an shrupt steepening to the High Central Plateau from 800 feet. (240 meters) to an 1100 feet (330 meters) level. Although ice stagmated in this zone; an evidenced by the ridged ablation moraine and eskers, several minor stages of recessional moraines near Rarney's Brook and Little Sandy Pond suggest earlier active flow. The features are probably related to the Kitty's Brook - Chain Lakes recessional-solation moraines and a topographically controlled flow from the High central Plateau. Growsse fillings in the area post-date the tidged ablation moraine and may be considered a funal constructional form in the sequence of deglaciation.

On the lower Halls May-Gull Fond level, the northeasterly trending valley trains darived much of their material from this stage and final deglaciation. No climatic causes are envisaged for the stillstand, rather it is assumed that the event was a result of ice clearing the lowland and leaving the higher plateau ice as an independent body.

Final ice within the field area stagmated in situ just south of the Gaff Topsail, as denoted by hummocky disintegration moraine and a zone of thick drift. Moltwater from this stagmant zone and ice remaining on the hills west of the Kitty's Brook - Chain Lakes trough cut large'channels in the value system and constructed the postglacial dolta noted in Chaoter IV.

The plateau just south of the field area and north of Hinds Lake may be considered as one of the last ice areas in Newfoundland. Another was probably located on the Buchan's Plateau further south, from which ice flowed radially (Murray, 1955).

No date has been obtained for the two final glacial events in the field area (positions 4 and 5). Andrews (1972) describes average marginal recessions of 860 feet (260 meters) per year between 12,000 and 7,000 B.P. for south and northwest margins of the Laurentide ice sheet. For the northeast corner he applies a retreat value of 66 feet (20 meters) per year. The northeast coast of Newfoundland was probably not affected by warn air masses similar to those at the continental ice margin, rather, the cold Labrador current was a more likely influence. Ice marginal positions for Newfoundland as speculated by Prest (1970) take account of this. Thus, Andrews' northeastern values are most applicable to conditions on the northeast coast of the faind.

Retreat rates of 66 feet (20 meters) per year and a distance of 19 miles (30 kms.) from the const'to the edge of the plateau (position 4, Figure 6-1) yield a value of 1500 years for deglaciation conditions. Re-applying the method to the distance from the coast to Gaff Topssil, (34 miles, 55 kms.), a time span of 2750 years is calculated for ice retreat to reach the area, though final ice probably remained at this position for some considerable time. If radiocarbon date, G.S.C.-1733 is used for constal deglaciation (12,000 ± 220 B.P.) then it would be expected that deglaciation of the area around Gaff Topsall occurred at about 9,250 B.P.

Grant (1969) dates the Ten Mile Lake readvance on the Northern Peninsula at 10,900 B.P., so it is logical that ice remained on the south High Central Plateau at this time. The dates calculated by Andrews method are speculative, however it seems that a more exact chronology will not be obtained in the immediate future. Macpherson (1973) suggests reasons for a considerable lag in postglacial vegetative colonization in Newfoundland which night rule out the possibility of accurately dating non-marine inlend events in the field area.

Further Research -

The field area and zones peripheral to it offer much scope for future glacial investigations. Note detailed research is required on the reconstinual-solution moraines in the Kitty's Brook - Chain Lakes area. Analysis of the mortheastern end of the valley complex near Sheffield Lake would further help in interpreting the deposite, since the present study includes only a small area at the southwest and of Chain Lakes.

Ice retreat in Indian Brook Valley and the probable lacustrine deposits related to this stage of deglaciation should be given further

107

attention. If may be that final topographic flow into the valley was . from the plateau areas to the north and south and that residual ice remained in the upper part of the valley for a considerable time after ice retreated from the coast.

2. A major glacial study should be undertaken directly to the east of the field area in the Twin Lakes - Botwood vicinity (Figure 6-2). An analysis of the Botwood delta complex and glaciofluvial deposite along the coset further north might provide additional data to the inner-outer drift zones theory.

Lundqvist (1965) has supported the Jenness (1960) argument by constructing a "supposed line of retardation" from just east of Bishops Falls on the Exploits River to Kings Point in Green Bay. It appears however that the boundary should be much further northeast than its present position immediately north of the Twin Lakes area. The writer has noted raised glaciofluvial terraces on the northeast coast of Notre Bars Bay, which may be further evidence of ice retardation sub-paralleling the coast.

Grant (1972) describes an easter of flow in the Twin Lakes-Notre Dame Eay area. In addition to the north to northeasterly flow, he suggests radial flow from a remnant ice mass south of Frozen Ocean Lake similar to the last ice centers of the Topsails and Suchans Plateau. Dated samples from the Botwood complex or the more northerly raised glaciomarine deposits would add creatence to the possibility that the lower north cossi of Newfoundland from Green Eay to Bonavists Bay was deglaciated at approximately the same time; an idea that cannot as web & substantisted.

108



3. Expansion of the field south, to the Buchans Plateau would be a worthwhile venture in terms of describing mother possible last ice center. It has been concluded that within the field area the "Caff Topsafis" was an area of last ice. Additionally, certainly the higher Buchans Plateau just southwest and northeast of Hinds Lake might be considered a last ice refuge since it is from this point that the terrain starts to slope away to the south and west. A study of the plateau with emphasis on the lloyd's Rivet valley, the Little Grand Lake valley and the lever Hinds Lake Plateau, would fill a large void in the glacial history of the island.

In the field of glacial geomorphology, the island of Newfoundland is one of the least studied areas in Ganada. With increasing demands being placed on environmental understanding derived from such research, it is hoped that further studies on the island will receive the financial and logistical support which they

deserve.

# SELECTED BIBLIOGRAPHY

ANDREWS, J. T. 1963

The cross-valley moraines of north-central Baffin Island, Northwest Territorias. Geog. Bull. Can. Vol. 8, No. 2, pp. 174-193.

ANDREWS, J. T. and KING, C. A. M. 1968

Comparative till fabrics and till fabric variability in a till sheet and a drumlin: a small scale study.

ANDREWS, J. T. and SMITH, D. I. 1970

Statistical analysis of till fabric. - methodology, local and regional variability.

Quart. J. Geol. Soc. London Vol. 125, No. 500, Part 4, pp. 503-542.

ANDREWS, J. T. 1970

A geomorphical study of post-glacial uplift with particular reference to Arctic Canada. 1.B.G. Special Publication #2. Institute of British Geographers. 156 p. London.

ANDREWS, J. T. 1972

The Wisconsin Laurentide Ice Shett - Despersal, Centers, Problems of Rates of Retreat and Climatic Implications. Paper read at the I.G.U./INQUA meeting Montreal.

BAIRD, D. M. 1931

Geology of the Burlington Peninsula. G.S.C. Paper, 21.

BOUSFIELD, E. L. 1960

Canadian Atlantic Sea Shells. National Museum of Canada, Cat. No. R 92 - 1659. Ottawa, 72 p. BROOKES, I. A. 1969

Late Pleistocene glacial events in southwest Newfoundland. Paper presented at INQUA Congress, Sect. 6, Group 2 Paris,

BROOKES, I. A. 1969

Late glacial-marine overlap in western Newfoundland. Can. J. Earth Sci. Vol. 6, pp. 1397-1404.

BRÜCKNER, W. D. 1969

Post-glacial features in Newfoundland, eastern Canada. Eclogae Geological Helvetiae. Vol. 62, No. 2.

CARROLL, D. 1970

Clay minerals: A guide to their X-ray identification. G.S.A. Special Paper. 126, pp. 45-55.

CHAMBERLAIN, T. C. 1895

Notes on the Geology of Newfoundland. Bull. Geol. Soc. Am. Vol. 6, p. 467.

COLEMAN, A. D. 1926

The Pleistocene of Newfoundland. J. of Geol. Vol. 34, pp. 193-223.

COOK, J. H. 1946

Kame complexes and perforation deposits. Am. J. Sci. Vol. 244, pp. 573-583.

COOPER, J. R. 1936

Geology of the south half of Bay of "Islands igneous complex, Newfoundland.

Nfld. Dept. Nat. Res. Geol. Sec. Bull. 4.

COWAN, W. R. (1968)

Ribbed moraine: till fabric analysis and origin. Can. J. Earth Sci. Vol. 5, No. 5, pp. 1145-1159.

DALY, R. A. 1921

Post-glacial warping of Newfoundland and Nova Scotla. Am. J. Sci. 20 Series 5, pp. 381-391. DAMMAN, A. W. 1964

Some forest types of central Newfoundland and their relation to environmental factors. Forest Science Monograph. Pub. of the Soc. of Am. Foresters.

·113

DOORNKAMP, J. C. and KING, C. A. M. 1971

Numerical Analysis in Geomorphology: an introduction. Edward Arnold Ltd., London. 372 p.

DREIMANIS, A. and VAGNERS, U. A. 1971

Bimodal distribution of rock and mineral fragments in basal til1. In Till, a Symposium. Goldthwaite, R. P. '(ed.) Columbus, pp. 237-250.

DYCK. W. and FYLES. J. G. 1963

Geological Survey of Canada Radiocarbon dates I and II. G.S.C. Paper. 63-21, pp. 17-18.

DYKE, A. S. '1973

A geomorphological map and description of an emerged Pleistocene delta, Eastport Peninsula, Newfoundland. Maritime Sediments: Vol. 8, No. 2, pp. 68-72.

EMBLETON, C. and KING, C.A.M. 1970

Glacial and Periglacial Geomorphology. Macmillan of Canada, Toronto, 608 p.

FAIRBRIDGE, R. W. 1958

Dating the latest movement of Quaternary sea level. Trans. N. Y. Acad. Sci. Vol. 20, p. 471.

FARRAND, W. G. and GAJDA, R. T. 1962

Isobases of the Wisconsin marine limit in Canada. Geog. Bull. No. 17, pp. 5-22.

.FLINT, R. F. 1928

Eskers and crevasse fillings. Am. J. Sci. Vol. 215, pp. 410-416. FLINT, R. F. 1940

Late Quaternary changes of level in western and southern Newfoundland.

114

Bull. Geol. Soc. Am. Vol. 51, pp. 1757-1780.

FLINT, R. F. 1951

Highland centres of former glacial outflow in northern North . America. Bull. Geol. Soc. Am. Vol. 62, pp. 21-38.

FLINT. R. F. 1971

Glacial and Quaternary Geology. John Wiley and Sons Inc., Toronto. 892 p.

FOLK, R. L. 1968

Petrology of Sedimentary Rocks. Hemphills Book Store, Austin, Texas. 170 p.

FRIEDMAN, G. M. 1961

Destinction between dune, beach and river sands from their textural characteristics. J. Sed. Pet. Vol. 31, pp. 514-529.

GERMAN, R. 1964

Korngrössen-Untersuchungen an glazigen und glazifluvialen Sedimenten Neues. Jahrb. Geol. u. Palaont. Vol. 7, pp. 388-390.

GLEN, J. W., DONNER, J. J. and WEST, R. G. (1957)

On the mechanism by which stones in till become oriented. Am. J. Sci. Vol. 255, pp. 194-205.

GODWIN, H. SUGGATE, R, P. and WILLIS, E. H. 1958

Radiocarbon dating of the eustatic rise in ocean level. Nature, Vol. 181, pp. 1518-1519.

GRANT, D. R. 1969

Late Pleistocene re-advance of piedmont glaciers in western Next ound land . Maritime Sediments. Vol. 5, No. 3, pp. 126-128.

# GRANT, D. R. 1969

Surficial deposits, geomorphic features, and late Quaternary history of the terminus of the Northern Peninsula of Newfoundland and adjacent Quebec-Labrador.

Maritime Sediments. Vol. 5, No. 3, pp. 123-125.

GRANT, D. R. 1970

Quaternary Geology, Great Northern Peninsula Island of Newfoundland. <u>G.S.C. Paper</u>. 70-1, Part A, pp. 172-174.

'GRANT, D. R. 1972

Surficial Geology, Western Newfoundland. G.S.C. Paper. 72-1, Part A, pp. 157-160.

GRANT, D. R. 1973

Canada-Newfoundland and Labrador Mineral Development Program. Project 6.

G.S.C. Paper. 73-1, Part A, pp. 196-198.

GRAVENOR, C. P. and KUPSCH, W. O. 1959

Ice disintegration features in western Canada. J. Geol. Vol. 67, pp. 48-64.

HARE, F. K. 1952

The climate of the island of Newfoundland, A Geographical Analysis. Geog. Bull, Vol. 2, pp. 36-88.

HARRIS, S. A. 1969

The meaning of till fabrics. in <u>Processes</u> and <u>Mathod</u> in <u>Canadian Geography - Geomorphology</u>. pp. 143-164. Mathuen Pub., Toronto.

HENDERSON, E. P. 1960

Surficial Geology, St. John's, Newfoundland. G.S.C. Map. 35-1959.

HOLMES, A. 1947 Kames,

Am. J. Sci. Vol. 245, pp. 240-249.

# HOPPE, G. 1948

The retreat of ice from the lower regions of Norrbetten as illustrated by the land formations. Geographics (Upsails). No. 20, pp. 104-108.

HOPPE, G. 1952

Hummocky moraine regions with special reference to the interior of Norrbotten. Geog. Annal. Vol. 34, pp. 1-71.

JENNESS, S. E. 1960

Late Pleistocene glaciation of eastern Newfoundland. Bull. Geol. Soc. Am. Vol. 71, pp. 161-180

KALLIOKOSKI, J. 1953.

Preliminary Map, Springdale, Newfoundland. G. S. C. Paper. 53-5.

KERR, J. H. 1870

Ice Marks in Newfoundland. Quart. J. Geol. Soc. London. Vol. 26, pp. 704-705.

LEE, H. A. 1962

Method deglaciation, age of submergence and rate of uplift west and east of Hudson Bay. Bull. Pertglacjainy No. 11, pp. 239-245.

LEE, H. A. 1965

'Investigation of eskers for mineral exploration. G.S.C. Paper. 65-14.

LEWIS, W. V. 1949

An esker in process of formation: Böverbreen, Jotunheimen, 1947. J. Glac. Vol. 1, No. 6, pp. 314-319.

LUNDQVIST, J. 1965

Glacial Geology in Northern Newfoundland. Geologiska Foreningensi. Vol. 87, pp. 285-306.

MACCLINTOCK, P. and TWENHOFEL, W. 1940.

Wisconson glaciation in Newfoundland. Bull. Geol. Soc. Am. Vol. 51, pp. 1729-1756.

MACCLINTOCK, P. and DREIMANIS, A. 1964.

Re-orientation of till fabric by overriding glacier in the St. Lawrence Valley. Am. J. Sci. Vol. 262, pp. 133-142.

#### MACLEAN, H. S. 1947

Geology and Mineral Deposits of the Little Bay Area. Nfld. Geol. Surv. Bull. 22.

MACPHERSON, J. B. 1973

Contribution to Discussion: Symposium on Retreat of Wisconsin Laurentide Ice Sheet. Arctic and Alpine Research.

MILNE, J. 1874

Notes on the physical features and the mineral deposits of Newfoundland. Quart J. Geol. Soc. Vol. 30, p. 726.

.

MURRAY, R. C. 1955

Directions of glacier ice motion in south-central Newfoundland. J. Geol. Vol. 63, pp. 268-274.

NEALE, E. R. W. and NASH; W. A. 1963

Sandy Lake Map Area. G.S.C. Paper. 62-28.

NEALE, E. R. W.

King's Point area Newfoundland, Topography and Glaciology. Unpublished Manuscript.

POWERS, M. C. 1953

A new roundness scale for Sedimentary Particles. J. Sed. Pet. Vol. 23, pp. 117-119.

PREST. V. K .. GRANT, D. R. and RAMPTON, V. N. 1967

Glacial Map of Canada. G.S.C. Map. 1253A.

PREST, V. K. 2968

Nomeficiature of moraines and ice flow features as applied to the Glacial Map of Canada. <u>G.S.C. Paper.</u> 67-57.

# PREST, V. K. 1970

Pleistocene Geology and Surficial Deposits. in <u>Geology and Economic Minerals of Canada</u>, ed. by Douglas, R. J. W. pp. 675-764. Queens Printer, Ottawa,

PRICE, R. J. 1966

Eskers near the Casement Glacier, Alaska. Geog. Annel. Vol. 48A, pp. 111-125.

RICHARDS, H. G. 1940

Marine Fleistocene fossils from Newfoundland. Bull. Geof. Soc. Am. Vol. 51, pp. 1781-1788.

ROGERSON, R. J. and TUCKER, C. M. 1972

Observations on the glacial history of the Avalon Peninsula. Maritime Sediments. Vol. 8, pp. 25-31.

ROSE, E. R., SANFORD, B. V. and HACQUEBARD, P. A.: 1970

Economic Minerals of Southeastern Canada. in <u>Geology and Economic Minerals of Canada</u>. ed. by Bouglas, R. J. W. pp. 305-364. Oweens Frinter, Ottawa.

RYDER, J. M. 1971

Some aspects of the morphometry of paraglacial alluvial fans in south-central B.C. Can. J. Rarth Sci. Vol. 8, No. 10, pp. 1252-1264.

RUTTER, N. W. 1969

Comparison of moralnes formed by surging and normal glaciers. Can. J. Earth Sci. Vol. 6, pp. 991-999.

SHEPARD, F. P. 1961

Sea level rise during the past 20,000 years. Zeits, für Geom. supplementband 3, pp. 30-35.

SLATT, R. M. 1972

Texture and composition of till derived from parent rocks of contrasting texture, southeast Newfoundland. Sed. Geol. Vol. 7, pp. 283-290. SMALLEY, I.J. and UNWIN, D. J. 1968

The formation and shape of drumlins and their distribution and orientation in drumlin fields. J. Glac. Vol. 7, pp. 377-390.

STALKER, A. MacS. (1960)

Surficial geology of the Red Deer - Stattler map area. G.S.C. Mem. 306.

STOKES, J. C. 1958

An esker like ridge in process of formation, Flatisen, Norway. G. Glac. Vol. 3, pp. 286-290.

SWIFT, P. J. P., SCHUBEL, J. R. and SHELDON, R. W. 1972 .

Size analysis of fine grained suspended sediments. J. Sed. Pet. Vol. 42, pp. 122-134.

TANNER, V. 1940

The Glaciation of Long Range of Western Newfoundland, A brief Contribution. Geol. Fören. Förhandl. Bd. 62, HA 4, pp. 361-368.

TERASMAE, J. 1963

Three C1, dated pollen diagrams from Newfoundland, Canada. Advancing Frontiers of Flant Sciences Vol. 16, pp. 149-162.

THOMAS, R. D., EDWARDS, D. W. A. and FULTON, R. J. 1971.

Legend explanation. G.S.C. open file reports 29, 59, 106.

TWENHOFEL, J. 1963

Physiography of Newfoundland. Am. J. Sci. Vol. 3, pp. 1-24.

TWENHOFEL, W. H. and MACCLINTOCK, P. 1940

Surface of Newfoundland. Bull. Geol. Soc. Am. Vol. 51, pp. 1665-1728.

TWENHOFEL, W. H. 1947

The Silurian of eastern Newfoundland with some data relating to physiography and Wisconsin Glaciation of Newfoundland. <u>Ad. J. Sci.</u> Vol. 245, pp. 65-122.

# WALCOIT, R. I. 1972

Late Quaternary vertical movements in eastern North America; quantitative evidence of glacio-isostatic rebound, <u>Reviews of Geophysics and Space Physics</u>. Vol. 10, No. 4, pp. 843-884.

WIDMER, K. 1950

Geology of the Hermitage Bay area Newfoundland. Unpublished Ph.D. thesis, Princeton University.

WILLIAMS, H. 1962

Botwood (West Half) Map Area Newfoundland. G.S.C. Paper, 62-9.

WILTON, W. C. 1957

Experimental Area No. 2, Halls Bay, Newfoundland Establishment Report. Dept. N. Affairs and Nat. Res. pp. 7-16. Forcestry Branch. Project N.F. 25.

#### APPENDIX 1

# TECHNIQUES, INSTRUMENTS AND ANALYSES

Various field and Laboratory methods were used in preparation of the thesis. While most are standard techniques, a list is provided with descriptions where necessary.

# Sampling Techniques

Unconsolidated drift subjected to the various analyses was obtained by a standard procedure. Generally, 5 pound (2.5 kgms.) esemples were acquired from cuts and packaged unsieved in polyethylene sample bags. Where good cuts were not available, pits were dug to a depth of 2-2.5 feet (0.62-0.76 meter) to avoid zones of weathering.

Silt samples obtained from a large deposit near Indian Brook (Chapter II) were sliced from a sedimentary section that had been removed intact from the field area and mounted on a metal retainer. Other slits analysed were procured in a memoer similar to the drift samples.

# Field Instruments

Various pieces of apparatus were in use throughout the field , season. In addition to the standard field equipment, a killace and fierman barometric altimeter was used. The instrument was zeroed at high tide level and field readings were interpolated to the nearest 2 feet (0.6 meters). Corrections for atmospheric temperature variations were applied using a sling psychrometer and a graphic scale provided with the altimeter. Most constel elevations and those within two hours walking distance of datum were obtained by this method.

A hand level was used to determine heights above the general surface of remote features and deposits.

# Till Fabric Analysis

Till fabric analysis was attempted on several deposits in the Kitty's Brook - Chain Lakes area. Each fabric analysis consists of ordentation and dip measurements which were plotted on polar equidistant projections (Chapter IV, Figure 4-4), with orientations on a series of radii spaced at 10-degree intervals and dips on concentric circles indicating the angle of slope from the horizontal. The circles were also spaced at 10 degree intervals. Data was measured to the measurest 5 degrees and all bearings were corrected for magnetic declifiction. Pebble size was generally from 1-2 inches (2.5-5 cms.) with a long to short axis ratio of not Iess than 2.511.

A measure of probability as to whether the clasts sampled ware preferentially oriented or not was calculated automatically in the field using Harris' (1969) method based on the conventional thi square test, in which the number of pebbles in a specified class is compared with the number of pebbles there would be if all the classes were equally filled. By following the graphic results one is able to determine exactly how many pebbles are required in the primary uble out of a total number counted to schieve a 95% level of confidence or the minimum significant orientetion count (M.S.O.C.).

To calculate the mean and standard deviation of the fabrics without the negating effect of a 360 degree diagram, points included

122

in an arbitrarily decided minor half of the projection were rotated 180 degrees. The means and standard deviations were calculated for a range of 0-180 degrees using a 20 degrees class interval. The results were then converted to azimuthal bearings relative to true north.

# Sizé Analysis

A dry sleve malysis was completed on all sediment samples. An upper limit of -3% (phi) and lower limit of 46 were used at whole \$\$ intervals. Each 100 gms. (3.5 or.) sample was oven dried and then sleved for 15 minutes on a Ro-tap vibrator. It is acknowledged that whole \$\$ units are too large for detailed sediment work, however the objective was to present a general quantitative description of various features and not a detailed sedimentological analysis.

Several samples containing large mounts of silt and clay were analysed for textural variations using a method of pipette analysis outlined by Polk (1966). The technique is based on the settling' valocities of particles computed with reference to Stokes Law,  $V = \frac{2}{9}$  (ds-dw)  $\frac{2}{m}$  r<sup>2</sup>, where: V = fall velocity in cms/sec., ds = densityof the sphere, dw = density of the fluid, g = acceleration due togravity, n = visconity of the fluid and r = the radius of the sphere.Folk re-arranged the equation and provided a constant "A", dependentupong, ds, and n, which is function of temperature. By knowingthe water temperature and particle density, the time required for anyparticle to fall any depth through a column of water can be calculated,and thus, the amount of a specific size of sedement contained in a . given sample.

Prior to pipette analysis, the sodiment samples were wet sieved and the material finer than 40 was treated with H<sub>2</sub>O<sub>2</sub> overnight to remove organic impurities. Samples were dispersed with .1N sodium hexametaphosphate (Calgon) and the analysis completed as outlined by

Folk.

From the various textural analyses, cumulative and individual percentage weights were calculated for the samples and the results plotted on Cumulative (arithmetic ordinate) Curves and Frequency Curves. The following statistical parameters were determined from Folk (1968):

Mean = ME = <u>\$16 + \$50 + \$84</u>

Inclusive Graphic Standard Deviation =  $\sigma_1 = \frac{\phi 84}{4} - \frac{\phi 16}{4} + \frac{\phi 95 - \phi 5}{6.6}$ 

Inclusive Graphic =  $Sk = \frac{\phi_{16} + \phi_{84} - 2\phi_{50}}{2(\phi_{84} - \phi_{16})} + \frac{\phi_{5} + \phi_{95} - 2\phi_{50}}{2(\phi_{95} - \phi_{5})}$ 

Kurtosis =  $K_{G} = \frac{\phi 95 - \phi 5}{2.44 (\phi 75 - \phi 25)}$ 

X-ray Analysis

To test the warfactions in composition of certain samples an X-ray analysis of the socions finer than 84 was attempted (outlined by Cargoll, 1970). Oriented samples were prepared for analysis by costing glass slides with supensions of the material to be tested. Three slides of each emple were made. They word air dried and the process was repeated to assure a thick coating. The procedure allowed the elay particles to active this he out face parallel to the slide, the orientation necessary for scanning.

An initial X-ray pattern using a Fhillips diffractometer and a proportional counter was run from 2-30 degrees 20 (theta) at the following instrument settings: 2 degrees 20 scanning.apeed, 40 KV., 20 MA., TC = 4, 200 c.p.s. and a chart speed of 20 mm. per minute. The results in degrees 20 were converted to d- spacings in angebrons and the clay minerals present were determined. To test for montmorificonte, a second slide of each sample was treated overnight in a dessicator with ethylene glycol at a temperature of 80°C. As soon as possible after removing the slide from the dessicator a pattern. was run from 2-15 degrees 20 st the settings used previously. A final slide of the samples was analysed to determine whether kaolinite or chlorite was present. The samples were first placed in a preheated muffle furnace at 600°C for one hour. Upon removal they were scanned at the above settings.

#### Stereo Microscope Analysis

A stereo microscope was used to analyse the 20 fraction of all sediment samples. From the study, particle roundness and sphericity were described using Powers (1953) criteria. The degree of polish on the quartz fractions was also noted and a search was made of specific samples for formaminifers and distoms.

125

# APPENDIX 2

# TEXTURAL ANALYSES COMPOSITES

The purpose of the following appendix is to present graphical composites of all samples analysed, plotted in various statistically descriptive combinations.

Rean size and skewness are plotted in Figure A-1. A good separation is obtained batween the silt samples (s) from Indian Brook valley and the esker (s) and till (t) samples. One esker sample is grouped with the tills. Although near symmetrical, its mean size does not vary substantially from the other esker samples, which are fine skewed.

In plotting skewness against standard deviation, Figure A-2, a well defined separation occurs between the sediment types. The till samples have a brond standard deviation which is a result of the glacially derived poorly sorted texture. Esker samples are sorted to a greater degree and generally finer skewed than the till; the silts although not as well sorted as the seker samples are strongly fine skewed.

A poor definition of water laid samples was obtained when skewness was plotted with kurtosis. From the results obtained it would be difficult to describe a specific hydrological environment. The tills are well separated from the esker and silt samples, since all water laid material was finely skewed.

It would be premature to explain the variances more than superficially based on so small a sample number. With a greater quantity of samples; boundaries should be defined more exactly, though

deviations from the norm would still be present.
# MEAN VERSUS SKEWNESS

S Units

28

### DEVIATION NDARI VERSU: SKEWNES

Inits Figure

ø

## KEWNESS VERSUS KURTOSIS

50-

.35-.30-.25-.20ŝ

BX.

### APPENDIX 3

### ABBREVIATIONS

C.N.R.

D. M. A. R.

D.R.E.E. M.S.O.C.

### Canadian National Railways

Department of Mines, Agriculture and Resources, Government of Newfoundland and Labrador.

Department of Regional Economic Expansion, Government of Canada.

Mean standard orientation count.

National Air Photo Library, Government of Canada.

N.A.P.. T.C.H.

Trans-Canada Highway.









